Do Financial Constraints Really Matter? A Case of Understudied African Firms^{*}

Michael Machokoto[†]

June 19, 2020

Abstract

Using a system of equations to account for the simultaneity, intertemporal and interdependent nature of corporate decisions, we document several new insights into how emerging market firms allocate funds across competing uses-of-funds. Emerging market firms save most of the operating cash flow. When the firms spend, they allocate the remainder to dividend payments first, followed by debt retirements, then equity repurchases, and lastly investments. This pecking order of prioritising savings and dividends ahead of other uses-of-funds highlight difficulties in accessing external finance and a stubbornly resilient signalling motive for firms operating under a high degree of information asymmetry and agency costs. We further find significant asymmetry and heterogeneity in the allocation of funds conditional on credit constraints, deviations from target, and around the financial crisis. Our findings signal the need for policies that improve access to external finance and information disclosure in emerging markets.

Keywords: Financial constraints, emerging markets, cash flow allocations, financial crisis, asymmetry in corporate decisions. **JEL classification:** G01; G31; G32

^{*}We thank Mustafa Caglayan, Siko Sikochi, Daniel Gyimah, Nadeem Aftab, seminar participants at the University of Northampton and Heriot-Watt University, two anonymous referees, and Mark P. Taylor (The Editor) for their helpful comments.

[†]The Faculty of Business and Law, University of Northampton, Waterside Campus, University Drive, Northampton, NN1 5PH, UK, **Tel:**+44 160 489 3484 **Email:**michael.machokoto@northampton.ac.uk

1 Introduction

Do financial constraints affect real economic activities? If so, what are the channels through which they affect real activities? How do we measure and test for financial constraints? These questions have been explored several times, but in all cases, the results are inconclusive.¹ However, apart from the evidence on developed economies, we know little about how financial constraints affect investment and financing decisions in emerging capital markets. We attempt to fill this lacuna by examining how emerging market firms allocate operating cash flow to savings, investments, dividends, debt retirements and equity repurchases. By examining all uses-of-funds as opposed to adopting a piecemeal approach of focusing on either investment-cash flow sensitivity or cash flow-sensitivity of cash, our study provides new insights on the impact of financial constraints on investment and financing decisions.

Changes in how firms allocate funds have real implications on firm-growth, employment and economic growth. For example, building-up cash reserves entail reducing or postponing current investments. At the same time, this accumulation of savings indicates increasing difficulties in accessing external finance, which is of interest to policymakers, particularly in less-developed capital markets. Similarly, a high allocation of funds to dividends signals a lack of other viable monitoring mechanisms in the form governance and institutional structures (Brav et al., 2005; Iturriaga and Crisóstomo, 2010; He et al., 2017) or the lack of further growth opportunities. The former requires that managers adopt a dividend payout to signal the quality of the firm in the presence of a high degree of information asymmetry and agency costs. At the same time, the latter is plausible, but less so for emerging markets that are less saturated and have lower barriers to entry (see

¹See, Allayannis and Mozumdar (2004), Almeida and Campello (2007), Andrén and Jankensgård (2015), Ascioglu et al. (2008), Chen and Chen (2012), Fazzari et al. (1988), Lewellen and Lewellen (2016), Hovakimian et al. (2004), Hu and Schiantarelli (1998), Kaplan and Zingales (1997) and Moyen (2007).

Panibratov, 2017). This rules out the latter and leaves the signalling motive as the most plausible reason why emerging market firms would payout most of the funds rather than retain and re-investment, which negatively impact on firm-growth and economic growth. On the other hand, high allocations to debt repayments and equity repurchases *per se* are indicative of better access to external finance, hence, boosting economic growth.² Therefore, understanding how emerging market firms allocate funds represents an interesting research question with wider economic implications.

To accomplish the above objective, we estimate a system of equations relating the five uses-of-funds to operating cash flow and several control variables and lagged uses-of-funds to account for the interdependence and intertemporal nature of corporate decisions. This aspect is particularly pertinent in emerging markets where access to external finance is much limited, and firms rely mostly on internal sources of capital, thereby, making investment and financing decisions more intertemporal and interdependent. We then test for asymmetry in cash flow allocations or sensitivities conditional on four commonly used measures of credit constraints. To shed more insights on whether cash flow sensitivities are asymmetric, we use a *quasi-natural experiment* in the form of the financial crisis, which is mainly external and reliably orthogonal to local credit market conditions in emerging markets. Using this unexpected external credit supply shock increases our ability to test and discriminate among several plausible propositions that have been advanced in the literature (see Almeida et al., 2004; Gatchev et al., 2010; Chang et al., 2014; Lewellen and Lewellen, 2016). In addition, we also investigate whether non-linearities in cash flow and deviations from target cash holdings, investment, dividends and capital structure influence allocations across the five uses-of-funds. This part of our analysis seeks to shed further empirical insights on the often-overlooked interdependence of investment

²There is a rich literature showing that access to finance has a positive effect on innovation and economic growth (see Levine, 1997; Beck and Levine, 2004; Beck et al., 2009; Brown et al., 2013; Acharya and Xu, 2017).

and financing decisions.

Using a sample of 5,940 firm-year observations from eight emerging economies from 2000-2015, and a system of equations framework for five uses-of-funds, we find a noteworthy pecking order in the allocation of funds and a significant interdependence of investment and financing decisions. Specifically, the sampled emerging market firms save 44% of operating cash flow. When they spend, they allocate the remainder in order of proportions to dividend payments first (19%), followed by debt retirements (15%), then equity repurchases (14%), and lastly investments (8%). This pecking order in the allocation of funds reveals several insights into investment and financing decisions in the presence of binding credit constraints, information asymmetry and agency costs. No-tably, the disproportionately high savings emphasise the maintenance or enhancement of financial flexibility as a more critical goal that ranks ahead of all other uses-of-funds when access to external finance is limited. Therefore, this accumulation of cash reserves, which often entails cutting back or postponing investments, should be of concern to investors and policymakers alike as it hampers firm-growth, and consequently, employment and economic growth in emerging markets.

Dividend payments, which consistently rank just below savings, appear sticky-down, implying that on average firms increase or maintain rather than decrease the payouts during the financial crisis. This finding of a sticky-down pattern in dividends is surprising and unique to our sampled non-utility and non-financial firms as it is in contrast with results in prior literature. For example, Shirai (2004) argues that firms can easily cutback on dividends to avert bankruptcy as they are not mandatory like interest and debt repayments. A notable exception is Floyd et al. (2015) who document similar resilience in dividend payouts during the financial crisis but only for US banks which use dividends to signal their profitability and solvency to stakeholders. We attribute the unique sticky-down pattern for the sampled non-utility and non-financial firms to the prevalence of

information asymmetry and agency costs in emerging markets that drive a stubbornly strong signalling motive against deteriorating business fundamentals during the financial crisis. The aforementioned stylised patterns, which are in stark contrast to those in the US where firms allocate 36% of operating cash flow to savings, 32% to debt retirements, 23% to investments, 9% to equity repurchases and only 1% to dividends as reported by Chang et al. (2014), highlight the utilitarian nature of the emerging market context as an independent sample for reconciling and generalising findings from developed economies.

Our findings of uniquely low investment-cash flow sensitivities (cash flow allocations to investments), even for constrained firms during the financial crisis, show the poor performance of this commonly used measure of credit or financial constraints in environments and around events where the constraints are supposedly more binding. Our results show that investment-cash flow sensitivities are more linked to under and over-investment problems than financial constraints *per se*, as firms with low (high) sensitivities underinvest (over-invest). This is in contrast with a strong correlation between investment and credit constraints in prior literature (see Fazzari et al., 1988; Biddle and Hilary, 2006; Almeida and Campello, 2007; Beatty et al., 2010). From our further analyses, cash flow sensitivity of cash, which until recently has received limited coverage, emerges as a more reliable and informative measure of credit constraints that correlates significantly with credit market conditions. These findings signal the need for a shift in research focus as firms are increasingly investing in intangible capital and cutting-back on physical or tangible investments.

The allocations of funds to debt retirements and equity repurchases for our sampled emerging market firms are comparatively lower and higher, respectively, than those in the US (see Chang et al., 2014). This difference indicates the less-developed nature of the capital markets in emerging economies. The sampled firms have approximately 85.4%– 86.2% of total assets financed by equity (high equity-dependence) as the emerging bond or debt markets are still in their infancy (see Shirai, 2004; Mu et al., 2013). Further, our results show significant interdependence in investment and financing decisions, which suggests that overlooking this critical aspect of investment and financing decisions could lead to biased inferences on cash flow sensitivities.

To support the robustness of our findings, we use the financial crisis as a *quasi-natural experiment* to examine how managers allocate funds across competing uses-of-funds. Again, we find that our main story holds as the firms in our sampled non-utility and non-financial firms save most of the operating cash flow and prioritise dividends ahead of other uses-of-funds. In additional analyses, we find that deviating from the target cash holdings, investments, dividends and capital structure has real implications on how firms allocate funds. Our results further suggest that firms above (below) the target cash holdings tend to build-up more savings and allocate less of the funds to investments, dividends, debt retirements and equity repurchases. For the above target-investment firms, we find the opposite as they save less, which further confirms the trade-off between savings and investments in markets where access to external finance is limited. Also, we find that above-target dividend firms allocate less funds to other uses-of-funds and that dividends are resilient or stick-down for our sampled non-utility and non-financial firms. This finding points to a strong signalling motive in emerging markets beleaguered by a prevalence of information asymmetry and agency costs, and signals the need for policies that improve information disclosure and access to external finance.

Our study makes several contributions to the literature. First, we provide further empirical evidence from a richer framework that simultaneously models all uses-of-funds while accounting for the intertemporal and independent nature of investment and financing decisions. Adopting this approach helps to introduce some order in the rich body of empirical evidence as it is in stark contrast to the literature using a *ceteris paribus* approach (piecemeal approach) of focusing on a single use of funds (e.g. Almeida et al., 2004; Chen et al., 2012; Guariglia and Yang, 2016; Moshirian et al., 2017). Our results not only confirm the significant interdependence in investment and financing decisions but also show that firms simultaneously allocate funds across savings, investments, dividend payments, debt retirements and equity repurchases (sources-equal-uses-of-funds). In addition, we find that deviating from optimal investment and financing levels have real implications on corporate decisions or outcomes. These findings suggest that estimates of cash flow sensitivities based on piecemeal approaches and static models should be interpreted with caution as they could lead to biased inferences. Second, we show that firms in emerging markets prioritise savings ahead of investments and other uses-offunds, which highlights limited access to external finance. Third, we show that usefulness of investment-cash flow sensitivity has declined and that the measure is more linked to under and over-investment problems rather than financial constraints. A substitute in the form of cash flow sensitivity of cash emerges as a more relevant measure of credit constraints that correlate with changes in credit markets. Fourth, our further analyses show that the significant asymmetries in cash flow sensitivity of cash reported in the literature disappear once negative-cash flow firms are excluded, this helps shed more light on the mixed empirical findings in the literature (Almeida et al., 2004; Riddick and Whited, 2009; Bao et al., 2012; Machokoto and Areneke, 2020). Fifth, we provide new evidence on the stickiness of dividend payouts in emerging markets even during the financial crisis when business fundamentals deteriorated. We attribute the dividend stickiness to the prevalence of agency costs and information asymmetry in less-developed capital markets. Sixth, we report a battery of robustness tests aimed at addressing mis-measurement errors associated with Tobin's q (a proxy of future growth opportunities) and provide meaningful comparisons on the relative performance of the methods in a new and unique context. Finally, we extend the analyses of financing constraints on all uses-of-funds to understudied emerging markets are institutionally different from advanced economies, where

the literature in concentrated. In doing so, we provide new empirical evidence from an independent and unique sample that helps to generalise and reconcile mixed findings in the literature.

The rest of the paper is organised as follows. Section 2 presents a brief background of the context and hypotheses. Section 3 introduces the empirical approach and describes the data used. Section 4 presents the empirical findings and robustness tests. Section 5 concludes.

2 The emerging-market context and hypotheses

In the following sub-sections, we present a brief overview of the relevant contextual issues to this study and the motivations of our hypotheses.

2.1 The emerging market context

By way of motivation, we plot two commonly used measures of financial development (stock market capitalisation-to-GDP and private debt-to-GDP) for the sampled countries, and other selected emerging and developed economies. Figure 1 presents a scatterplot of the average stock market capitalisation-to-GDP (%) and private credit-to-GDP (%) with the superimposed average GDP (constant 2010 USD) over the sample period.

PLEASE INSERT FIGURE 1 HERE

Figure 1 shows that most African countries consistently rank below other emerging and developed economies, except for South Africa. The scatterplots are in line with Mu et al. (2013), who find that the African bond markets are still in their infancy. Similarly, Gwatidzo and Ojah (2014) document a significant influence of non-traditional factors (e.g. education of the managers, location, legal infrastructure and location) on financing decisions of firms operating in less-developed African economies. The case of South Africa, as depicted in Figure 1, is unique along several dimensions. The country appears to have robust stock and bond markets as it is ranked second only to Switzerland based on stock market capitalisation-to-GDP, and fourth just below the US, Japan and UK based on private debt-to-GDP. Nevertheless, South Africa has lower levels of GDP per capita relative to the other selected exemplary emerging and developed economies. This finding highlights significant disparities, where finance is only accessible to a few companies and individuals. Our observation corroborates Gwatidzo and Ojah (2014) who find that, even within South Africa, only a few companies can access formal non-bank financing as most companies either cannot afford the interest rates charged by banks or fall-short of the stringent loan requirements imposed by lenders in risky capital markets (most companies and individuals do not have the required collateral or credit history).

The low levels of financial development depicted in Figure 1 for the sampled countries have several implications on how firms allocate funds. For example, Oztekin (2015) report significant differences in corporate debt levels, with South African firms (the only African country in their sample) having 13% of their total assets financed by debt, which is comparatively lower than those in Brazil (27%), Canada (18%), France (23%), Germany (16%), India (28%), Japan (23%), Singapore (19%), Switzerland (24%), UK (17%) and USA (23%). This limited access to debt, in particular long-term debt, increases equitydependence as reported by Mu et al. (2013), thereby making the payment of dividends a priority for most emerging market firms. The equity-dependence reinforces the dual role of dividends – as both monitoring (disciplinary role) and signalling devices – in markets characterised by a high degree of information asymmetry and agency costs (see Brav et al., 2005; Iturriaga and Crisóstomo, 2010; He et al., 2017). For the case of emerging markets, the disciplinary role of corporate debt is forfeited, and dividends naturally emerge as the best available and most viable option given that directly policing or monitoring managers is not only costly but fraught with institutional deficiencies that cannot be easily addressed by investors. The lack of a robust corporate debt market also increases reliance on internal financing sources such as retained earnings and cash reserves (see Guariglia and Yang, 2018), which implies a significant focus on buffering of cash reserves to hedge against future shortfalls. However, this conservatism (accumulation of large cash reserves) leads to under-investment problems as firms can only increase cash reserves by reducing or postponing current investments.

On the other hand, as access to external finance is limited in emerging markets, the correlation between investment and cash flow (investment-cash flow sensitivity) should be much higher as firms rely mostly on internal capital and are still heavily invested in tangible or physical capital as noted by Moshirian et al. (2017). In addition, as most of the corporate borrowings have short maturities and are in the form of bank loans (see Gwatidzo and Ojah, 2014; Sorge et al., 2017), firms are likely to commit a significant proportion of funds to debt retirements. This concentration of short-term borrowings not only increases maturity-mismatch and refinancing risks but could lead to short-termism, with managers focusing more on servicing and refinancing debt at the expense of other strategic or long-term goals.

In summary, a combination of the above unique peculiarities makes emerging markets a utilitarian context or an independent sample that is akin to a laboratory setting for reconciling and validating existing theories and findings from developed economies.

2.2 Hypotheses

A large body of literature starting with Fazzari et al. (1988) examines the effects of financial constraints on real decisions by examining the sensitivity of investment to cash flow. Firms facing binding financial constraints tend to rely mostly on internal funds as they have limited access to external financing sources (see, Almeida and Campello, 2007; Chen and Chen, 2012; Fazzari et al., 1988; Moshirian et al., 2017). The limited access to external finance should lead to high investment-cash flow sensitivity and cash flow sensitivity of cash, with the sensitivity being more pronounced in emerging economies that have less developed institutions relative to other advanced economies. Also, according to Moshirian et al. (2017), firms in emerging economies are structurally different from those in advanced economies as they operate with more physical capital. These high investments in physical capital should lead to a higher correlation between investment and operating cash flow for emerging market firms relative to those reported in developed economies.³

At the same time, Almeida et al. (2004), Riddick and Whited (2009) and Bao et al. (2012) also show that cash flow sensitivity of cash increases with credit constraints. Similarly, Khurana et al. (2006) report a decrease in cash flow sensitivity of cash with financial development. In addition, Chang et al. (2014) also find that US firms allocate a high proportion of funds to savings and debt retirements. In contrast, they allocate less to investment, equity repurchases and dividends in that order. Lewellen and Lewellen (2016) documents similar variations in cash flow allocations for US firms, except that most of the funds are towards investments rather than savings. The studies mentioned above point to significant variations in the allocation of funds with credit constraints. Therefore, our first hypothesis is stated as follows:-

Hypothesis 1 (H1): Cash flow sensitivities for cash, dividends and debt retirements are higher and lower for investments and equity repurchases.

Although there is some emerging consensus that credit constraints affect corporate decisions, the measurement and channels or mechanism of this effect is debated. The mixed results across different measures of financial constraints have further compounded

³Investments in physical capital have declined significantly in advanced economies as more firms are increasingly concentrating on innovation or research and development (Brown et al., 2009; Brown and Petersen, 2009, 2015).

the debate. For example, Hadlock and Pierce (2010) and Chang et al. (2014) have both cast doubts on the appropriateness of the KZ Index (Kaplan and Zingales, 1997) as their test returning results that are impulsive and unreliable. Similarly, Chen and Chen (2012) find no differences in investment-cash flow sensitivity between supposedly constrained and unconstrained firms classified based on traditional measures of credit constraints (size, firm-age, credit-ratings, dividend payouts and corporate governance index). However, a recent study by Chang et al. (2014) finds significant differences in cash flow sensitivities based on some of the above measures of credit constraints. They find that constrained firms categorised based on the WW Index (Whited and Wu, 2006), HP Index (Hadlock and Pierce, 2010), size, dividend-paying status and credit ratings allocate most of their funds to savings and equity repurchases, and less to investments, dividends and debt retirements relative to their unconstrained counterparts. Lewellen and Lewellen (2016) report similar differences which are in stark contrast to Chen et al. (2012) who find no differences. They attribute the disparities in cash flow sensitivities, in particular, the low investment-cash flow sensitivity as reported by Chen et al. (2012), to the use of noisy measures or proxies of cash flow.

In addition to the debate on the measures of financial constraints, an emerging tranche of the literature on whether cash flow sensitivity of cash is asymmetric or not reports similarly mixed results. For example, whereas Almeida et al. (2004), Khurana et al. (2006), Chang et al. (2014), Lewellen and Lewellen (2016),Grullon et al. (2018) and McLean and Zhao (2018) find a positive effect of cash flow on changes in cash (the cash flow sensitivity of cash), Riddick and Whited (2009) and Bao et al. (2012) find this effect to be negative. They single out mis-measurement errors associated with Tobin's q as the main reason for the differences in cash flow sensitivity of cash. However, their proposed way of addressing the mis-measurement errors via generalised method of moments (GMM) estimators based on higher-order moments is similarly debated as Almeida and Campello (2007), Chang et al. (2014) and Lewellen and Lewellen (2016) show that the estimators return economically impulsive cash flow sensitivities of cash. This mixed evidence leaves the central question of whether financial constraints affect real decisions open to debate, especially in emerging markets where the literature is sparse, and access to external finance is limited. Accordingly, we propose and test the following hypothesis using several proxies of financial constraints:-

Hypothesis 2 (H2): Cash flow sensitivities are heterogeneous or asymmetric and differ across different measures of financial constraints.

3 Data and methodology

3.1 Data

We extract accounting data from *Datastream* and macroeconomic data from The World Bank database over the period 2000-2015. Our sample coverage is purely dictated by data availability. Following the standard convention in the literature, we drop firms in heavily regulated financial and utility sectors (Brav, 2009; Flannery and Rangan, 2006; Oztekin, 2015). We exclude firms with missing data on key variables. To reduce the compounding effect of outliers or merger and acquisitions, we drop firms with more than 100% growth in assets or sales and winsorise all variables used at the lower and upper one percentile. Our final sample consists of 639 firms with 5,940 firm-year observations from Egypt, Ghana, Ivory Coast, Kenya, Morocco, Nigeria, South Africa and Tunisia. All variables used are defined in Appendix A.

PLEASE INSERT TABLE 1 HERE

Table 1 reports the summary statistics and differences in cash flow and uses-of-funds conditional on four proxies of financial constraints. Panel A, for the main variables, shows that the mean (median) of 0.018 (0.008), 0.079 (0.064), 0.054 (0.034), 0.019 (0.000), -0.061 (-0.057), and 0.156 (0.138) for changes in cash (Δ Cash), investments (Capex), dividends (Div), changes in debt (Δ D), changes in equity (Δ E) and cash flow (CF), respectively. Panel B of Table 1 shows that, on average, constrained firms have lower cash flow (CF), investments (Capex), pay less in dividends (Div) and retire less debt (Δ D) and repurchase less equity (Δ E). At the same time, constrained firms save (Δ Cash) relatively more than unconstrained firms. These differences appear to be in line with our expectations that constrained firms are likely to prioritise enhancing financing flexibility (Δ Cash) ahead of other uses-of-funds (Capex, Div, Δ D and Δ E).

PLEASE INSERT TABLE 2 HERE

Table 2 presents the pairwise Spearman (Pearson) correlations in the above (below) diagonal. The pairwise correlations show that cash flow (CF) is positively correlated with changes in cash (Δ Cash), investments (Capex) and dividends (Div), while it is negatively correlated with changes in debt (Δ D) and changes in equity (Δ E). These correlations are in line with our initial predictions on the uses-of-funds (cash flow allocations). Table 2 also shows that changes in cash (Δ Cash) are negatively correlated with investments (Capex) and dividends (Div), which suggests that firms build-up cash reserves by forgoing investments and curtailing dividend payments. The correlations of the other control variables are consistent with the literature, and for brevity, we only further discuss key variables of interest.

3.2 Methodology

To examine cash flow sensitivities, we follow Gatchev et al. (2010) and simultaneously estimate the following system of equations:

$$\begin{bmatrix} \Delta \operatorname{Cash}_{ijt} \\ \operatorname{Capex}_{ijt} \\ \operatorname{Div}_{ijt} \\ \Delta \operatorname{D}_{ijt} \\ \Delta \operatorname{E}_{ijt} \end{bmatrix} = \mathbf{L} \begin{bmatrix} \operatorname{CF}_{ijt} \end{bmatrix} + \mathbf{K} \begin{bmatrix} \Delta \operatorname{Cash}_{ijt-1} \\ \operatorname{Capex}_{ijt-1} \\ \operatorname{Div}_{ijt-1} \\ \Delta \operatorname{D}_{ijt-1} \\ \Delta \operatorname{E}_{ijt-1} \end{bmatrix} + \mathbf{M} \begin{bmatrix} q_{ijt-1} \\ \operatorname{SG}_{iJt-1} \\ \operatorname{Size}_{ijt-1} \\ \operatorname{PPE}_{ijt-1} \end{bmatrix} + \left\{ \begin{bmatrix} \epsilon_{ijt}^{\Delta \operatorname{Cash}} \\ \epsilon_{ijt}^{\epsilon_{ijt}} \\ \epsilon_{ijt}^{\epsilon_{ijt}} \\ \epsilon_{ijt}^{\epsilon_{ijt}} \\ \epsilon_{ijt}^{\epsilon_{ijt}} \end{bmatrix} \right\}$$
(1)

where ΔCash_{ijt} , Capex_{ijt} , Div_{ijt} , ΔD_{ijt} and ΔE_{ijt} are the changes in cash, investments, dividends, changes in debt and changes in equity, respectively, for firm *i* in country *j* at time *t*; **L**, **K** and **M** are matrices of parameter coefficients of size 5×1 , 5×5 , 5×4 , respectively; CF_{ijt} is cash flow and the control variables are; q_{ijt-1} is lagged marketto-book value, SG_{iJt-1} is lagged sales growth, $Size_{ijt-1}$ is the lagged logarithm of total assets, and PPE_{ijt-1} is lagged property, plant and equipment; and, $\epsilon_{ijt}^{\Delta \text{Cash}}$, $\epsilon_{ijt}^{\text{Div}}$, $\epsilon_{ijt}^{\Delta \text{D}}$ and $\epsilon_{ijt}^{\Delta \text{E}}$ are the error terms. The literature informs the choice of control variables (e.g., Almeida et al., 2004; Chen and Chen, 2012; Gatchev et al., 2010; Chang et al., 2014). The sources-equal-uses-of-funds or adding-up constraint (cash flow should equal the uses-of-funds) requires that $\mathbf{i'L=1}$, $\mathbf{i'K=0}_{1\times 5}$ and $\mathbf{i'M=0}_{1\times 4}$ (see Gatchev et al., 2010). The adding-up constraint will naturally be satisfied if there are no income items that have been directly reported in total equity instead of the income statement (a practice commonly known as "dirty-surplus accounting" (see Chang et al., 2014; Lewellen and Lewellen, 2016)).

To study the impact of financial constraints on cash flow sensitivities, we split the sample based on the WW Index (WW) (Whited and Wu, 2006), HP Index (HP) (Hadlock and Pierce, 2010), firm-size (Size) and firm-age (LogAge).⁴ In each year for each country, we categorise firms as being constrained (unconstrained) if they are below (above) the median firm-size and firm-age, and unconstrained (constrained) if they are below (above) the median of the WW Index and HP Index. Other extant studies use different categorisation or classification schemes such as the upper and lower terciles or quantiles of the distribution to study asymmetry in cash flow sensitivities (see Almeida et al., 2004; Almeida and Philippon, 2007; Bao et al., 2012; Almeida et al., 2013; Chen et al., 2012). As using these schemes increase the likelihood of finding differences, we contend that using the median is a more conservative approach and preferable, especially in cases where the sample size is small, or the distribution of the data is skewed. We, however, take comfort in that untabulated results based on the upper and lower terciles or quantiles classification schemes do not materially differ from our main findings.

We estimate our models simultaneously using seemingly unrelated regressions (SUR) in line with recent literature (see Gatchev et al., 2009, 2010; Andres et al., 2014; Chang et al., 2014). Using this framework enables us to simultaneously account for both the interdependence and intertemporal nature of investment and financing decisions, and also the sources-equal-uses-of-funds or adding-up constraint (cash flow should equal the uses-of-funds). As argued by Gatchev et al. (2010), overlooking the intertemporal and interdependent nature of corporate investment and financing decisions could lead to bias inferences on cash flow sensitivities. However, for robustness and to ensure compara-

⁴We do not use the KZ Index (Kaplan and Zingales, 1997) as in prior studies given that our untabulated results appear to be impulsive and unreliable. In addition, Hadlock and Pierce (2010) show that the KZ Index is an unreliable proxy of financial constraints.

bility with prior studies, we also present results for the equation-by-equation estimates (separately estimated equations) based on several estimation techniques. Specifically, we also estimate our models using ordinary least squares (OLS), fixed effects (FE), the higher-order moments estimator of Erickson and Whited (2000, 2002) (GMM3–GMM5), instrumental variables 2SLS (IV-2SLS) (Baum et al., 2008), instrumental variables GMM (IV-GMM) (Baum et al., 2003, 2008), difference general method of moments (DIFF-GMM) (Arellano and Bond, 1991), system general method of moments (SYS-GMM) (Blundell and Bond, 1998) and Panel Vector Autoregression models (PVAR) (Abrigo and Love, 2016). The latter methods use higher-order moments or instrumental variables to address mis-measurement errors associated with Tobin's q, a proxy of future growth opportunities.⁵

4 Results

In this section, we first estimate a system of equations with and without the sourcesequal-uses-of-funds (adding-up) constraint to understand how firms allocate operating cash flow. Next, we examine the effects of financial constraints on cash flow sensitivities conditional on several commonly used proxies of financial constraints. We then use the financial crisis as a *quasi-natural experiment* to better understand this effect during significant contractions in credit supply. Finally, we present a battery of robustness tests aimed at addressing several problems surrounding the study of cash flow sensitivities.

4.1 The investment and financing-cash flow sensitivities

Table 3 presents the estimation results of a system of equations depicted by Equation (1) that relate the uses-of-funds to cash flow and several control variables. Columns (1)–(5)

⁵Our results are robust to mis-measurement errors associated with Tobin's q and using several alternative estimation techniques.

and (6)-(10) present estimation results for models without and with the sources-equaluses-of-funds (adding-up) constraint, respectively.

PLEASE INSERT TABLE 3 HERE

Columns (1)–(5) of Table 3, for estimates of cash flow sensitivities based on models without the sources-equal-uses-of-funds constraint, show that firms allocate most of the internally generated cash flow to savings (43.8%), followed by dividends (17.7%), debt (13.6%), equity (10.2%) and investments (8.4%) in that order. The results show that, on average, a firm increases savings by 2.97%, and reduce debt by 1.14%, while paying 1.13% in dividends, and at the same time allocating 0.87% to equity repurchases, and only investing 0.57% for a one standard deviation increase in operating cash flow. This pecking order in cash flow allocations is consistent with our first hypothesis. It shows that emerging market firms are subject to significant credit constraints as the allocations to savings are 5.2 times higher than those to investments (Capex). The results are in stark contrast to Chang et al. (2014), who find that US firms allocate 28% and 33% of operating cash flow to investments and savings, respectively.

Columns (1)–(5) further show that investment and financing decisions are intertemporal and interdependent as the coefficients of the lagged uses-of-funds (Δ Cash, Capex, Div, Δ D and Δ E) are significant. Our untabulated results, when we exclude the lagged uses-of-funds, further confirm the bias in the estimates of cash flow sensitivities based on models that overlook the intertemporal and interdependent nature of investment and financing decisions. The estimates from these models return comparably higher cash flow allocations to equity purchases and dividend payments and lower allocations to savings, investments and debt retirements. Our further analyses using an equation-by-equation approach confirms this bias and emphasise the need to account for the intertemporal and interdependent nature of corporate decisions.

The estimates of cash flow sensitivities based on models with the sources-equal-usesof-funds (adding-up) constraint, in Columns (6)-(10), are similar to those in Columns (1)-(5) based on models estimated without the adding-up constraint. This finding suggests that our results are robust to the critique that cash flow sensitivities which are not estimated simultaneously and without explicitly imposing the adding-up constraint lead to biased inferences (see Gatchev et al., 2010). Instead, our results corroborate Chang et al. (2014) and Lewellen and Lewellen (2016) who argue that if variables are consistently defined in the absence of "dirty-surplus accounting" (a practice of directly reporting income items in total equity rather than the income statement - this is akin to by-passing the income statement), the adding-up constraint will naturally be satisfied. However, in contrast to the aforementioned studies, Columns (6)-(10) show lower allocations to investments (Capex) and equity repurchases (ΔE), and higher allocations to savings, dividends and debt retirements (except for Chang et al. (2014) who report cash flow allocations of 32% to debt retirements in the US). The differences indicate the cautious investment approaches of firms that operate in environments where access to external finance is limited. At the same time, enhancing financial flexibility by building substantial cash reserves and signalling to the market by pre-committing to pay dividends appear to be more pressing goals for firms in emerging markets relative to those in developed economies.

In summary, our estimates of cash flow sensitivities reveal several noteworthy patterns; (1) the high allocations to savings and dividends suggest two primary motives – the need to enhance financial flexibility through buffering cash reserves when access to external finance is limited and at the same pre-committing to paying dividends as a way of reducing information asymmetry and agency costs, and (2) a high reliance on short-term debt or short-term debt dependence as evidenced by the significantly higher cash flow allocations to debt retirements, (3) low-equity repurchases (which indicate equity dependence) highlighting the less-developed nature of the capital markets, and (4) under-investment problems as evidenced by the significantly lower allocation of funds to investments. Although the above results reveal unique insights into how firms allocate operating cash flow, they are limiting as the linear models used implicitly assume homogeneity in cash flow sensitivities. Yet, theory and anecdotal evidence point to significant asymmetry or heterogeneity in investment and financing decisions.

4.2 The effects of financial constraints on cash flow sensitivities

We next explore the impact of financial constraints on investment and financing decisions by comparing cash flow sensitivities or allocations between constrained and unconstrained firms. We categorise or classify firms into the low (high) regime if they are below (above) the median of the WW Index (WW), HP Index (HP), firm-size (Size) and firm-age (LogAge) in each year for each country. Table 4 summarises the results for the subsamples.

PLEASE INSERT TABLE 4 HERE

Column (1) of Table 4 shows that constrained firms (high-WW, high-HP, small and young firms) save 50%–60% more than their unconstrained counterparts (low-WW, low-HP, large and mature firms). The asymmetric cash flow allocations to savings conditional on financial constraints are in line with our second hypothesis (Hypothesis 2) and consistent but higher than those reported by Chang et al. (2014) in the US (of between 35% and 38%). This propensity to save as popularised by Almeida et al. (2004), Riddick and Whited (2009) and Bao et al. (2012), which is higher in our case of emerging market firms, indicate the primacy of maintaining or enhancing financial flexibility through holding substantial cash reserves. This finding is line with Almeida et al. (2004) and suggests that holding vast cash reserves is particularly important when access to external finance

is likely to be more uncertain given the firm's current financial position (as would be the case for constrained firms) and its operating environment. As argued by Guariglia and Yang (2018) for the case of Chinese firms, firms operating in emerging markets that are characterised by limited access to external finance tend to rely mostly on self-financing sources (retained earnings and cash holdings). The need to hedge against future shortfalls explains the high propensity to save that we document in an environment beleaguered by institutional voids.

As shown in Column (2), our sampled firms only allocate 8.4% of the funds to investments. This allocation is much lower than expected and those reported by Chang et al. (2014) and Lewellen and Lewellen (2016) of 23% and 26%, respectively. The differences are somewhat surprising as emerging market firms are still heavily invested in physical capital and have limited access to external finance (Moshirian et al., 2017), which should result in higher investment-cash flow sensitivities (allocations of cash flow to investments) relative to developed economies. To the extent that emerging markets offer a unique and independent sample, our contrasting findings further corroborate Chen et al. (2012) who report decreases in investment-cash flow sensitivities are not good proxies for financial constraints as they are lower rather than higher for emerging market firms that are more subject to binding credit constraints.

Column (3) shows that allocations of funds to dividend payments are similarly asymmetric and consistently lower for financially constrained relative to unconstrained firms. These results are in line with the asymmetric savings and investment behaviour we observed in Columns (1) and (2), which suggest that financially constrained firms prioritise enhancing financial flexibility ahead of investments and would similarly, as in Column (3), not pay much in dividends. The findings prevail despite the central role of dividends as signalling devices for firms operating in emerging markets that are characterised by a high degree of information asymmetry. We put forth two reasons why the sampled firms would allocate lower proportions of operating cash flow to dividend payments: (1) constrained firms being less-profitable have less to payout and would not pre-commit to dividends they cannot sustain, and (2) the few profitable firms would instead save rather than spend as future income-flows are highly uncertain. This conservatism arises due to the considerable wedge between internal and external costs of funds in emerging markets which reinforces the propensity to save rather than spend. Thus, only unconstrained firms with better prospects or future growth opportunities allocate a significant portion of operating cash flow to dividend payments as a way of signalling their quality, and in the process, reduce information asymmetry and improve access to external finance.

Next, we explore the cash flow allocations to debt retirements and equity repurchases for which the relevant literature is sparse, especially in emerging economies with lessdeveloped capital markets. Columns (4) and (5) show mixed evidence across the four proxies of financial constraints as we find that financially constrained (unconstrained) firms based on the WW Index and firms-age (HP Index and firm-size) allocate a higher (lower) proportion of operating cash flow to debt retirements. We find similarly mixed results on equity repurchases, in Column (5), with firms identified as unconstrained (constrained) based on the WW Index and firms-age (HP Index and firm-size) repurchasing more (less) equity than constrained (unconstrained) firms. In this instance, it is not clear why the results based on the WW Index and firms-age (LogAge) categorisation or classification schemes are opposite those based on HP Index and firm-size (Size). This situation obtains despite the significant positive Spearman (Pearson) correlation of 0.862 (0.821) between the WW and HP Index in Table 2. The mixed results not only highlight the difficulties encountered when attempting to measure or study the impact of credit constraints on corporate decisions, but also the need for further theoretical frameworks or models. These extensions would inform the empiricist about the underlying channels or mechanisms through which credit constraints affect real decisions.

Overall, our results suggest that financial constraints significantly affect the allocation of funds and that the propensity to save (maintaining or enhancing financial flexibility) positively correlates with credit constraints. At the same time, binding financial constraints are more likely to result in lower rather than higher investment-cash flow sensitivity. This observation is new and important to the literature as it helps reconcile the mixed empirical findings on investment-cash flow sensitivity. We argue that using a system of equations offer a better framework to study the impact of financial constraints, especially, in emerging markets where the intertemporal and independent nature of investment and financing decisions is more apparent with limited access to external finance.

4.3 The effects of financial constraints on cash flow sensitivities through the financial crisis

Building on the results in the previous section, we next use the financial crisis as a *quasi-natural experiment* to examine whether cash flow sensitivities vary with credit constraints before and during the 2008–09 contractions in credit supply. Table 5 summarises the results for the pre-crisis (Before) and crisis (After) periods.

PLEASE INSERT TABLE 5 HERE

Table 5 reveals several noteworthy changes in cash flow sensitivities around the financial crisis. The average sampled firm appears not to alter its savings around the financial crisis, but instead, significantly increase cash flow allocations to investments and dividend payments. The increased allocations to investments, investment-cash flow sensitivity, is as expected and in line with the mainstream literature which finds that, when faced with binding credit constraints, firms increasingly rely on internal financing sources (operating cash flow) (see Fazzari et al., 2000; Guariglia and Yang, 2016). However, this increase in the correlation between investments and cash flow is inconsistent with Chen and Chen (2012) and Machokoto et al. (2019) who document a marked decrease in investment-cash flow sensitivity in the US and UK, respectively, for both constrained and unconstrained firms around the financial crisis. Our results differ from the two aforementioned studies because emerging market firms are still heavily invested in physical or tangible capital and heavily reliant on internal capital sources, especially during contractions in credit supply, as the capital markets are comparatively less-developed.

Columns (3) and (8), for all firms, show a 26% increase in cash flow allocations to dividend payments from 0.142 in the pre-crisis period to 0.179 during the financial crisis. This increase is significant at 1% level and surprising as firms had to contend with binding credit constraints during the financial crisis (as evidenced by an 8% decrease in cash flow for our sample firms). At the same time, we also find a 6% and 44% curtailment in cash flow allocations to debt retirements and equity repurchases, respectively. However, the decrease is only significant for equity repurchases and not debt retirements, which similarly shows the over-reliance on equity finance in emerging markets. Our untabulated results further show that corporate debt marginally increased from 13.8% to 14.6%over the crisis-period, which explains why we observe an insignificant decrease in debt retirements. This equity dependence, which ranges between 85.4%–86.2% of total assets, is synonymous with the less-developed nature of emerging capital markets (see Mu et al., 2013). Put differently, the insignificant changes in cash flow allocations to debt retirements could also point to difficulties in servicing debt, which is likely to be more problematic in emerging markets where most of the borrowings are in the form of bank loans with short maturities.⁶

⁶Sorge et al. (2017) find that short-term debt constitutes as high as 51%, 63%, 42%, 78% and 49% of corporate debt in Brazil, Russia, India, China and South Africa, respectively. Similarly, Booth et al. (2001) report ranges of 24% to 76% in short-term debt across ten developing countries over the period 1980–1991. They further find that most of the short-term debt is in the form of bank loans. In addition,

Next, we study the differences in cash flow allocations between constrained and unconstrained firms around the financial crisis. As the financial crisis was primarily an exogenous credit supply shock that originated in the US sub-prime mortgage crisis, we contend that this set-up resembles a *quasi-natural experiment* where the credit supply shock is reliably orthogonal to local investment and financing opportunities (see Chari et al., 2008; Popov and Rocholl, 2018). Hence, any changes that we observe or document around the financial crisis, in Table 5, are less likely to result from other confounding or feedback effects.

The analyses around the financial crisis reveal several stylised changes and heterogeneity in cash flow allocations. Table 5 shows that unconstrained firms significantly increased savings while their constrained counterparts reduced allocations to savings. The decrease in savings for credit-constrained firms mirrors the decline in cash flow during the financial crisis and point to a diversion of funds toward protecting or smoothing investments. For the changes in cash flow allocations to investments, we only find a significant increase for unconstrained firms with their constrained counterparts experiencing an insignificant or muted change. This finding is in line with Table 4, and further shows that investment-cash flow sensitivity is increasingly becoming an unreliable measure of credit constraints.

Our sampled non-utility and non-financial firms significantly increased dividends during the financial crisis, except for large firms that can more easily dispense with the need to signal their quality using dividends. In line with the results in Tables 1 and 4, the increased allocation of funds to dividend payments for the other firm sub-groups (excluding large firms) highlights the central role of dividends as monitoring and signalling devices in emerging markets. As emerging markets are characterised by a high degree of informa-

Gwatidzo and Ojah (2014) find that firms in Africa prefer bank loans to non-bank debt as the former is availed with less-collateral and mostly based on long-standing relationships. They further find that non-bank debt is scarce and where available, it is costly and often accompanied by restrictive covenants.

tion asymmetry and agency costs, dividends are the only available and viable monitoring and signalling devices given that debt is inaccessible and directly policing managers is fraught with institutional deficiencies. These unique aspects of emerging markets are behind the stickiness and resilience in dividend payouts we have so far documented. The finding is similar to Floyd et al. (2015) who document significant resilience in dividends around the financial crisis in the US but only for banks that use the payouts to signal their profitability and solvency to key stakeholders. For our sampled firms, which are non-utility and non-financial firms, dividends assume a dual role of signalling to investors and disciplining managers by preventing them from misusing free-cash-flow given that governance structures in emerging markets are less-developed.

For the dynamics in cash flow allocations to debt retirements, we find similarly mixed and inconclusive results as those we tabulated in the previous section across different proxies of financial constraints. These findings indicate that the existing measures of credit constraints do not always lead to the same conclusions in different contexts, which calls for the development of context-specific proxies. Our final set of results show significant and consistent decreases in funds allocated to equity repurchases, with the reductions being more pronounced for unconstrained firms that are less-equity dependent and have better access to capital markets. This finding is in line with our expectations and Wesson et al. (2015) who find similarly low levels of equity repurchases which were only allowed much later on in South Africa (from 1 July 1999 onwards). Our results suggest that the emerging share repurchases market is still in its infancy owing to several institutional deficiencies. The deficiencies take the form of rigid announcement requirements, noncancellation of own shares repurchased, and inconsistencies in both the tax treatment and application of regulatory rules as noted by Wesson et al. (2015) in the exemplary case of South Africa that dominates our sampled countries.

Taken together, our analyses around the 2008–09 credit supply shock suggest that

credit constraints have a significant effect on both investment and financing decisions in emerging markets. Our results further show the increasing unreliability of investmentcash flow sensitivity as a measure of financial constraints, with the cash flow sensitivity of cash (the propensity to save) emerging as a more reliable proxy of credit constraints that correlates closely with underlying or prevailing credit market conditions.

4.4 Deviations from target and non-linearities in cash flow sensitivities

In this final part of our study, we examine non-linearities in cash flow sensitivities and how deviations from the target – that is being below or above the median lagged cash holdings, investments, dividends, debt, equity capital and cash flow – affect the allocation of funds. Table 6 summarises the estimation results for our additional analyses.

PLEASE INSERT TABLE 6 HERE

Columns (1)–(5) of Table 6 show that firms with cash holdings (Cash) above (below) the median seem to build-up more (less) savings, allocate less (more) to investments, dividends, debt retirements and equity repurchases. This accumulation of cash reserves as shown by the lower allocations to investments in Column (2) entails cutting back or postponing current investments, which are critical for firm-growth, and consequently, employment and economic growth in emerging markets. For sub-samples based on investments (Capex), we find that firms below (above) the median allocate relatively more (less) funds to savings and debt retirements. In contrast, they allocate less (more) to investments, dividends and equity repurchases. This allocation of funds is in line with Table 4 and suggests that firms under-invest due to binding credit constraints. As our previous results show, Columns (1)–(5) (for below-target investment firms) also suggest that constrained firms attempt to hedge against future shortfalls by increasing savings and further cutting-back on current investments. On the other hand, firms that overinvest (above-target investment firms) save less, pay more dividends and have higher investment-cash flow sensitivity, which are all features associated with unconstrained firms rather than constrained ones as popularised in the literature (see Fazzari et al., 1988; Biddle and Hilary, 2006; Beatty et al., 2010). These differences further buttress our earlier findings that cash flow sensitivity cash (investment-cash flow sensitivity) is becoming a more (less) reliable proxy of credit constraints.

Table 6 also shows that firms with above-median dividend payments allocate fewer funds to other uses-of-funds, except for dividend payments. This finding, which is further supported by the decrease in dividends with firm-growth, suggests that dividends are important and relatively sticky in the presence of a high degree of information asymmetry and agency costs. This finding is in line with several other studies documenting a significant effect of capital market development on corporate financing decisions (see Brown et al., 2013; Sorge et al., 2017). We further find that deviating from target debt significantly affects investment and financing decisions as firms with above (below) target debt allocate relatively more (less) to other uses-of-funds, except for savings. This way of allocating funds further perpetuates the under-invest problem that we observed for constrained firms in Table 4, especially within the Africa context where most of the corporate borrowings are in bank loans with shorter maturities (see Gwatidzo and Ojah, 2014).

On the other hand, the high-investments for above target-debt firms could signal over-investment issues associated with the conflict of interest between shareholders and creditors (agency problems) as noted by Khémiri and Noubbigh (2019). We also find that equity-dependent firms accumulate more cash reserves, invest less and pay more in debt. At the same time, they retire and repurchase less debt and equity, respectively.

Our further analyses based on cash flow in Columns (1)–(5) of Table 6 (Panel A),

which are motivated by the debate on whether cash flow sensitivity of cash is asymmetric (see Almeida et al., 2004; Riddick and Whited, 2009; Bao et al., 2012; Machokoto and Areneke, 2020), show that above-target firms allocate most funds to savings, investments and dividends. These firms also use some of the new funds to retire debt and repurchase equity. In Panel B, for our restricted sample of positive-cash flow firms (CF>0), we find significant asymmetries on dividends, debt retirements and equity repurchases, but not on savings and investments. This new finding, which is free from biases associated with ad-hoc or ex-ante sample splitting approaches in the literature, suggests that negative-cash flow firms mostly drive the asymmetry reported by Riddick and Whited (2009) and Bao et al. (2012) on cash flow sensitivity of cash as we do not find evidence of dis-savings even at very low levels of operating cash flow. The finding highlights a high propensity to save that does not appear wane with increases in operating cash flow against a backdrop of limited access to external finance.

To summarise, as our findings suggest that binding credit constraints affect investment and financing decisions, they signal the need to hasten the implementation of pro-capital market development policies in emerging markets.

4.5 Robustness

In this section, we implement a battery of robustness tests. First, we re-estimate our main models using several alternative techniques to facilitate comparisons with prior studies. Using different estimators enable us to gauge or assess the sensitivity of our results to mis-measurement errors related to Tobin's q (a proxy for future growth opportunities) that could bias our inferences (see Erickson and Whited, 2000, 2002; Riddick and Whited, 2009). Table 7 summarises the estimation results using several alternative techniques (for brevity, we only report the coefficients of cash flow and Tobin's q).

PLEASE INSERT TABLE 7 HERE

Our estimation results of the modified version of Equation (1), excluding the lagged independent variables, using an equation-by-equation approach via pooled ordinary least squares (OLS) and ordinary least squares with fixed effects (FE) appear reasonable and closer to satisfying the sources-equal-uses-of-funds constraint. The OLS and FE estimates show that firms in emerging markets, as exemplified by the eight sampled countries, have higher cash flow sensitivity of cash (Δ Cash) and cash flow sensitivity of dividends (Div). In comparison, they have lower investment-cash flow sensitivity (Capex). On overall, the estimates based on OLS and FE are consistent with our main findings, except for the cash flow sensitivity of changes in debt (Δ D) and equity (Δ E), which appear to be lower and higher than expected, respectively. However, as the equation-by-equation approach does not consider the intertemporal and interdependent nature of investment and financing decisions, the results are not entirely unexpected and should be interpreted with caution. This oversight could lead to biased inferences on cash flow sensitivities (see Gatchev et al., 2010; Chang et al., 2014).

Next, we discuss the estimates of cash flow sensitivities based on the higher-order moments estimator of Erickson and Whited (2000, 2002) (GMM3–GMM5) that corrects for potential mis-measurement errors associated with Tobin's q, a proxy of future growth opportunities. For this part of our analysis, we estimate a modified version of Equation (1) that excludes the lagged independent variables *via* GMM3–GMM5. Our estimation results show that τ , an index of the measurement quality for Tobin's q that varies between zero (poor proxy) and one (very good), ranges between 0.182 and 0.810. This range seems acceptable in our case. However, the estimates of the cash flow sensitivities based on GMM3–GMM5 appear to be economically implausible and in all cases violate the sources-equal-uses-of-funds or adding-up constraint (as the sum of the estimated cash flow sensitives exceeds one). This result highlights a possible limitation of the higherorder moments estimator of Erickson and Whited (2000, 2002), and corroborates Almeida et al. (2010), Chang et al. (2014) and Lewellen and Lewellen (2016) who similarly find the estimates to be imprecise in some cases.

We now turn our focus to estimates of cash flow sensitivities based on instrumental variables 2SLS (IV-2SLS) (Baum et al., 2008), instrumental variables GMM (IV-GMM) (Baum et al., 2003), difference general method of moments (DIFF-GMM) (Arellano and Bond, 1991), system general method of moments (SYS-GMM) (Blundell and Bond, 1998) and Panel Vector Autoregression models (PVAR) (Abrigo and Love, 2016). These dynamic panel data estimators have been shown to perform well in modelling the dynamic nature of corporate decisions, while at the same time addressing potentially endogeneity problems (see Dang, 2013; Flannery and Hankins, 2013). The estimators in our case use instruments to correct for potential mis-measurement errors associated with Tobin's For the PVAR models, we first time-demean the data and then purge the panel q. fixed effects using the forward orthogonal deviation or Helmert transformation. We first estimate a modified version of Equation (1) that excludes the lagged independent variables (lagged use-of-funds) using instrumental variables 2SLS (IV-2SLS) and instrumental variables GMM (IV-GMM). For our estimation results of Equation (1) via the difference GMM (DGMM) and system GMM (SGMM), we do not include the lagged independent variables of the other uses-of-funds. However, for the estimation results of Equation (1)via the Panel Vector Autoregression models (PVAR), we include all the lagged independent variables (for the five uses-of-funds). We use the second-to-third lags of the peer average Tobin's q (the peer average Tobin's q is calculated based on the four-digit SIC codes) as instruments for the IV-2SLS estimator and the second-to-third lags of Tobin's qas instruments for the IV-GMM estimator.⁷ To reduce over-identification issues or problems associated with instrument proliferation (too many instruments) (see Roodman, 2009), we restrict our instruments to the second-to-fourth lags for the difference GMM

⁷The peer-firm average Tobin's q is a suitable instrument as it is less noisy and correlates with GDP growth, a measure of long-term growth opportunities.

estimations (DGMM), the third-to-fourth lags for the system GMM (SGMM), and the second-to-third lags for Panel Vector Autoregression models (PVAR).

The validity of our instruments for the difference GMM and system GMM estimations is confirmed by both the Hansen (J) and second-order autocorrelation (m2) tests as they show no significant evidence of serial correlation. Table 7 further shows that the sums of the estimated cash flow sensitivities (Δ Cash+Capex+Div+ Δ D+ Δ E) based on difference GMM, system GMM and PVAR models are lower than one and closer to satisfying the adding-up constraint (\sum Uses_i = 1). In addition, and more importantly, the hierarchy or pecking order in the allocation of funds based the more efficient system GMM (SGMM) is in line with our main results and suggests that our sampled emerging market firms save most of the operating cash flow. When they spend, they allocate the remainder of the funds in order of proportions as followings – dividend payments first, followed by debt retirements, then equity repurchases, and lastly, investments (Capex). Based on the above additional findings from instrumental variable (IV) estimators, we conclude that our findings are robust to using alternative estimation techniques and potential mis-measurement errors associated with Tobin's q.

As a further robustness check, we also study the time-series variation in cash flow sensitivities. To accomplish this objective, we estimate a modified version of Equation (1) that excludes the lagged independent variables *via* seemingly unrelated regressions (SUR) and Fama and MacBeth (1973) two-step procedure (FM). The Fama and MacBeth (1973) two-step procedure is implemented as follows; (1) in the first step, cross-sectional regressions are estimated for each period, and (2) then in the second step, the coefficients from the first step are averaged to obtain the coefficients for the full sample period. According to Lewellen and Lewellen (2016), using annual cross-sectional regressions (FM) corrects for both time-series and cross-sectional dependence in firm-level datasets, while at the same time allowing for the relationship between cash flow and uses-of-funds to vary over time. Table 8 summarises the time-series estimates of cash flow sensitivities.

PLEASE INSERT TABLE 8 HERE

Table 8 shows significant time variations in cash flow sensitivities. Despite this significant variation over the sample period, the cash flow sensitivity of cash (savings) has remained high relative to other uses-of-funds and ranges from a low of 30.3% to a high of 68.2%. Consistent with our previous results, debt retirements appear to have almost disappeared around the financial crisis (2007–2009), and then, rebounded post-2009. Similarly, equity repurchases peaked in 2007 just before the onset of the financial crisis and decreased significantly thereafter. These changes are consistent with Table 5 and suggest that the financial crisis had a significant impact on how firms allocate funds, even in emerging markets such as Africa that are less-integrated with the US, the origin of the 2008–09 financial crisis. On overall, the seemingly unrelated regressions (SUR) and Fama and MacBeth (1973) two-step procedure (FM) return similar estimates of cash flow sensitivities, which further suggest that our results are robust to both time-variations, and using different model specifications and estimation techniques.

For the analysis based on vintage or period of listings, we sub-divide the sample into three 5-year sub-periods (namely; 2000–05 (L2000–05), 2006–10 (L2006–10), and 2011– 15 (L2011–15)) and categorise or classify our sampled firms based on the year of listing. Based on these sub-samples, we find significant differences in cash flow sensitivities, with firms listed post–2011 saving and investing relatively more than those listed in preceding sub-periods (L2000–05, L2006–10). Firms listed in the latter period (2011–15) not only appear to allocate less funds to dividends but also debt retirements and equity repurchases. These differences point to an increasing need to enhance financing flexibility in industries that are increasingly becoming concentrated (consolidated) as evidenced by the decline in new listings and rise in the untabulated Herfindahl-Hirschman Index (HHI) from a low of 4.5 to a peak of 10.3 over the sample period. Next, we examine the robustness of our results to alternative sub-sampling as our sample appears to be relatively heterogeneous in terms of geographic and industrial distribution. In Panel A of Table 9, we split the sample into two sub-groups; namely, South Africa and other countries.⁸ In Panel B, we split the sample into five industrial subgroupings; namely, Industrials (IND), Health Care (HC), Consumer Goods and Services (CG&S), Technology and Telecommunications (T&T) and Others (Basic Materials and Oil & Gas). Adopting this approach enables us to assess whether the cash flow sensitivities that we document vary across industries and between South Africa and other countries. Using this approach, in a way, addresses the uneven distribution of the sample, with South Africa that is comparatively more developed than the rest of the other countries dominating the sample. Table 9 summarises the estimation results for our sub-sample analyses.

PLEASE INSERT TABLE 9 HERE

Panel A of Table 9 shows that, relative to South African firms, firms in other African countries save similar proportions of operating cash flow (the cash flow sensitivity of cash as popularised by Almeida et al. (2004)). The similar cash flow sensitivities of cash (savings of 46%–47%) across the sub-country groupings are in line with our main results. They suggest that enhancing financial flexibility is of prime importance for firms operating in emerging markets characterised by institutional voids. Our estimates of savings between 46% and 47%, which are one and a half to three times higher than the 15% to 33% reported for US firms by Chang et al. (2014) and Lewellen and Lewellen (2016), emphasises the more central role of internal capital sources (such as retained earnings and cash reserves) in less-developed capital markets.

⁸As firm-year observations for each of the other countries are few, except for South Africa, we are unable to present and draw meaningful cross-country comparisons using our multi-equation research framework. We acknowledge this limitation in our study and point to the lack of rich datasets as one of the main reasons why there is a dearth of research in emerging markets (especially in Africa).

On the other hand, the significant differences in the other cash flow sensitivities (Capex, Div, ΔD and ΔE) between South Africa and other countries reflect the differences in stages of capital market development. For example, the higher investment-cash flow sensitivity (Capex) of 12.8% shows that credit constraints are more binding in Egypt, Ivory Coast, Kenya, Morocco, Nigeria and Tunisia relative to South Africa (with 5.7%). Similarly, South Africa appears to have a comparatively more active equity repurchases market, which again evidences a larger and more developed capital market. At the same time, the higher cash flow sensitivities of changes in debt ($\Delta D - 22.3\%$ for other countries vs 15.8% for South Africa) are due to the comparatively higher concentration of corporate debt with shorter maturities in other African countries (55% of total debt) relative to South Africa (44%). As most of the emerging market corporate debt is in the form of bank-loans with shorter maturities as reported by Sorge et al. (2017), the over-reliance on short-term debt further increases exposure to maturity mismatch and refinancing risks. This heavy reliance on short-term debt could be detrimental to firms in emerging markets as it leads to short-termism with managers focusing more on servicing and refinancing debt at the expense of other strategic or long-term goals.⁹

As a final robustness check, we examine the variations in cash flow allocations across broad industries or sectors. Panel B of Table 9 shows significant variations in cash flow allocations across our five broad industries.¹⁰ Firms in the Consumer Goods and Services (CG&S), Industrials (IND), and Technology and Telecommunications (T&T) sectors save more than those in other sectors as they allocate 51%–58% of their operating cash flow to savings. In line with our main findings, we observe that cash flow allocations to investments, investment-cash flow sensitivity, are consistently low across the five broad industries. This finding suggests that our main results are robust to sub-sampling by

⁹Appendix B shows that our results are robust to controlling for macroeconomic conditions and differences in the stages of economic development across countries.

¹⁰Appendix C shows similar variations in cash flow and uses-of-funds across industries.

industry or sector.

We further find that the high allocation of funds to dividends we have documented in the previous sections is mostly concentrated in Basic Materials and Oil & Gas sectors (Others) which dominate the emerging market corporate universe. Our cross-industrial analyses also reveal that sampled firms in the Technology and Telecommunications (T&T) and Others (Basic Materials and Oil & Gas) sectors allocate the least proportion of funds to debt retirements as 55%–63% of their borrowings are in the form of long-term debt. These allocations of funds are comparatively higher than those for sampled firms in other sectors that have between 48% and 54% of their borrowings as short-term debt. This high concentration of short-term debt exposes the sampled firms to maturity mismatch and refinancing risks, and account for the spike in cash flow allocations to debt retirements that we documented preciously around the financial crisis (Table 5). Next, we find that sampled firms in the Consumer Goods and Services (CG&S) and Industrials (IND) sectors allocate the least proportion of funds to equity repurchases (6%-7%). In contrast, those in other sectors are more equity dependent with allocations to equity repurchases that range from 25% to 30%. These cross-industrial variations are not unexpected and point to significant industrial heterogeneity in how emerging market firms use funds.¹¹

5 Conclusions

Using a large sample of firms from eight emerging economies over the period 2000-2015, and a system of equations that account for the overlooked intertemporal and interdependent nature of corporate decisions, we uncover several unique insights into how firms

¹¹Appendices D and E show that our results are not affected by the way we define the cash flow variable, a debated issued in the literature (see Chang et al., 2014; Lewellen and Lewellen, 2016). However, we also find the two other commonly used proxies of cash flow (CF1 – operating income *plus* depreciation-to-total assets and CF2 – net income before extraordinary items *plus* depreciation *minus* dividends-to-total assets) to be more volatile and less comprehensive, hence, our focus on the cash flow measure from the statement of cash flows rather than the statement of comprehensive income.

operating in these unique markets allocate funds to savings, investments, dividend payments, debt retirements and equity repurchases. We advance the literature by presenting new tests on asymmetry and non-linearities in cash flow sensitivities for all five uses-offunds.

First, we find that emerging firms allocate most of their operating cash flow to savings. When they spend, they allocate funds in order of proportions to dividend payments ahead of other uses-of-funds, followed by debt retirements, then equity repurchases, and finally, investments. This pecking order in cash flow allocations emphasises the importance of maintaining or enhancing financial flexibility when access to external finance is limited. Second, dividend payments, which consistently rank just below savings, appear to be sticky-down and vital as, on average, firms increase or maintain rather than curtail the payouts during the financial crisis. The pressing signalling motive mainly drives this stickiness in dividends against deteriorating business fundamentals in an environment characterised by high agency costs and information asymmetry. Third, cash flow allocations to investments, investment-cash flow sensitivity as popularised in the literature, are very low and appear to be poor measures of financial constraints. They are consistently lower rather than higher as would be expected for supposedly constrained firms, and only increase for unconstrained firms instead of the constrained ones during the financial crisis. Fourth, cash flow allocations to debt retirements and equity repurchases appear to be comparatively lower and higher than those in the US, respectively, which reflects the less-developed nature of emerging markets. Finally, we document evidence suggesting that corporate investment and financing decision are significantly intertemporal and interdependent, and that, if these peculiarities are overlooked as in the literature, could lead to biased inferences on cash flow sensitivities.

In general, our empirical findings, which offer a more holistic view of cash flow allocations in emerging markets, show that internal capital sources still predominate external ones. As our results show, this leads to the prioritisation of savings ahead of investments, which hampers firm-growth, and consequently, employment and economic growth. Our empirical analyses further reveal that investment-cash flow sensitivity is not a good measure financial constraints, even within the context of emerging markets where access to finance is limited, and during the financial crisis when credit constraints were more pronounced. Cash flow sensitivity of cash, which until recently has been overlooked in the literature, emerges as a more reliable and informative proxy of credit constraints that appears to correlate significantly with changes in capital markets. This signals the need for a shift in research focus as economies are transiting towards intangible capital that requires considerably higher levels of financial flexibility, which can take the form of spare borrowing capacity or cash reserves.

References

- Abrigo, M. R. M. and Love, I. (2016). Estimation of Panel Vector Autoregression in Stata. The Stata Journal, 16(3):778–804.
- Acharya, V. and Xu, Z. (2017). Financial dependence and innovation: The case of public versus private firms. *Journal of Financial Economics*, 124(2):223–243.
- Allayannis, G. and Mozumdar, A. (2004). The Impact of Negative Cash Flow and Influential Observations on Investment–Cash Flow Sensitivity Estimates. Journal of Banking & Finance, 28(5):901–930.
- Almeida, H. and Campello, M. (2007). Financial Constraints, Asset Tangibility, and Corporate Investment. *Review of Financial Studies*, 20(5):1429–1460.
- Almeida, H., Campello, M., and Galvao, A. F. (2010). Measurement Errors in Investment Equations. The Review of Financial Studies, 23(9):3279–3328.
- Almeida, H., Campello, M., and Weisbach, M. S. (2004). The Cash Flow Sensitivity of Cash. The Journal of Finance, 59(4):1777–1804.
- Almeida, H., Hsu, P.-H., and Li, D. (2013). When Less is More: Financial Constraints and Innovative Efficiency. SSRN Scholarly Paper ID 1831786, Social Science Research Network, Rochester, NY, USA.
- Almeida, H. and Philippon, T. (2007). The Risk-Adjusted Cost of Financial Distress. The Journal of Finance, 62(6):2557–2586.
- Andres, C., Cumming, D., Karabiber, T., and Schweizer, D. (2014). Do markets anticipate capital structure decisions? — Feedback effects in equity liquidity. *Journal of Corporate Finance*, 27:133–156.

- Andrén, N. and Jankensgård, H. (2015). Wall of cash: The investment-cash flow sensitivity when capital becomes abundant. *Journal of Banking & Finance*, 50:204–213.
- Arellano, M. and Bond, S. (1991). Some Tests of Specification for Panel Data: Monte Carlo Evidence and an Application to Employment Equations. *The Review of Economic Studies*, 58(2):277–297.
- Ascioglu, A., Hegde, S. P., and McDermott, J. B. (2008). Information asymmetry and investment–cash flow sensitivity. *Journal of Banking & Finance*, 32(6):1036–1048.
- Bao, D., Chan, K. C., and Zhang, W. (2012). Asymmetric cash flow sensitivity of cash holdings. *Journal of Corporate Finance*, 18(4):690–700.
- Baum, C. F., Schaffer, M. E., and Stillman, S. (2003). Instrumental Variables and GMM: Estimation and Testing:. The Stata Journal, 3(1):1–31.
- Baum, C. F., Schaffer, M. E., and Stillman, S. (2008). Enhanced Routines for Instrumental Variables/Generalized Method of Moments Estimation and Testing:. *The Stata Journal*, 7(4):465–506.
- Beatty, A., Liao, W. S., and Weber, J. (2010). The Effect of Private Information and Monitoring on the Role of Accounting Quality in Investment Decisions^{*}. *Contemporary Accounting Research*, 27(1):17–47. _eprint: https://onlinelibrary.wiley.com/doi/pdf/10.1111/j.1911-3846.2010.01000.x.
- Beck, T., Fuchs, M., and Uy, M. (2009). Finance in Africa Achievements and Challenges. Technical Report WPS5020, The World Bank.
- Beck, T. and Levine, R. (2004). Stock markets, banks, and growth: Panel evidence. Journal of Banking & Finance, 28(3):423–442.

- Biddle, G. C. and Hilary, G. (2006). Accounting Quality and Firm-Level Capital Investment. *The Accounting Review*, 81(5):963–982. Publisher: American Accounting Association.
- Blundell, R. and Bond, S. (1998). Initial Conditions and Moment Restrictions in Dynamic Panel Data Models. *Journal of Econometrics*, 87(1):115–143.
- Booth, L., Aivazian, V., Demirguc-Kunt, A., and Maksimovic, V. (2001). Capital Structures in Developing Countries. *The Journal of Finance*, 56(1):87–130.
- Brav, A., Graham, J. R., Harvey, C. R., and Michaely, R. (2005). Payout policy in the 21st century. *Journal of Financial Economics*, 77(3):483–527.
- Brav, O. (2009). Access to Capital, Capital Structure, and the Funding of the Firm. The Journal of Finance, 64(1):263–308.
- Brown, J. R., Fazzari, S. M., and Petersen, B. C. (2009). Financing Innovation and Growth: Cash Flow, External Equity, and the 1990s R&D Boom. *The Journal of Finance*, 64(1):151–185.
- Brown, J. R., Martinsson, G., and Petersen, B. C. (2013). Law, Stock Markets, and Innovation. The Journal of Finance, 68(4):1517–1549.
- Brown, J. R. and Petersen, B. C. (2009). Why has the Investment-Cash Flow Sensitivity Declined so Sharply? Rising R&D and Equity Market Developments. *Journal of Banking & Finance*, 33(5):971–984.
- Brown, J. R. and Petersen, B. C. (2015). Which investments do firms protect? Liquidity management and real adjustments when access to finance falls sharply. *Journal of Financial Intermediation*, 24(4):441–465.

- Chang, X., Dasgupta, S., Wong, G., and Yao, J. (2014). Cash-Flow Sensitivities and the Allocation of Internal Cash Flow. *The Review of Financial Studies*, 27(12):3628–3657.
- Chari, V. V., Christiano, L. J., and Kehoe, P. J. (2008). Facts and Myths About the Financial Crisis of 2008. Working Paper 666, Federal Reserve Bank of Minneapolis, Minneapolis, USA.
- Chen, H. J. and Chen, S. J. (2012). Investment-Cash Flow Sensitivity Cannot be a Good Measure of Financial Constraints: Evidence from the Time Series. *Journal of Financial Economics*, 103(2):393–410.
- Chen, Q., Chen, X., Schipper, K., Xu, Y., and Xue, J. (2012). The Sensitivity of Corporate Cash Holdings to Corporate Governance. *Review of Financial Studies*, 25(12):3610–3644.
- Dang, V. A. (2013). An Empirical Analysis of Zero-Leverage Firms: New Evidence from the UK. International Review of Financial Analysis, 30:189–202.
- Erickson, T. and Whited, T. M. (2000). Measurement Error and the Relationship between Investment and q. Journal of Political Economy, 108(5):1027–1057.
- Erickson, T. and Whited, T. M. (2002). Two-Step GMM Estimation of the Errors-in-Variables Model Using High-Order Moments. *Econometric Theory*, 18(3):776–799.
- Fama, E. F. and MacBeth, J. D. (1973). Risk, Return, and Equilibrium: Empirical Tests. Journal of Political Economy, 81(3):607–636.
- Fazzari, S. M., Hubbard, R. G., and Petersen, B. C. (1988). Financing Constraints and Corporate Investment. Brookings Papers on Economic Activity, 1(1):141–195.
- Fazzari, S. M., Hubbard, R. G., and Petersen, B. C. (2000). Investment-Cash Flow

Sensitivities are Useful: A Comment on Kaplan and Zingales. *The Quarterly Journal* of *Economics*, 115(2):695–705.

- Flannery, M. J. and Hankins, K. W. (2013). Estimating Dynamic Panel Models in Corporate Finance. Journal of Corporate Finance, 19:1–19.
- Flannery, M. J. and Rangan, K. P. (2006). Partial Adjustment Toward Target Capital Structures. Journal of Financial Economics, 79(3):469–506.
- Floyd, E., Li, N., and Skinner, D. J. (2015). Payout policy through the financial crisis: The growth of repurchases and the resilience of dividends. *Journal of Financial Economics*, 118(2):299–316.
- Gatchev, V. A., Pulvino, T., and Tarhan, V. (2010). The Interdependent and Intertemporal Nature of Financial Decisions: An Application to Cash Flow Sensitivities. *The Journal of Finance*, 65(2):725–763.
- Gatchev, V. A., Spindt, P. A., and Tarhan, V. (2009). How do firms finance their investments?: The relative importance of equity issuance and debt contracting costs. *Journal of Corporate Finance*, 15(2):179–195.
- Grullon, G., Larkin, Y., and Michaely, R. (2018). Are U.S. Industries Becoming More Concentrated? SSRN Scholarly Paper ID 2612047, Social Science Research Network, Rochester, NY.
- Guariglia, A. and Yang, J. (2016). A balancing act: Managing financial constraints and agency costs to minimize investment inefficiency in the Chinese market. *Journal of Corporate Finance*, 36(Supplement C):111–130.
- Guariglia, A. and Yang, J. (2018). Adjustment behavior of corporate cash holdings: the China experience. *The European Journal of Finance*, 24(16):1428–1452.

- Gwatidzo, T. and Ojah, K. (2014). Firms' Debt Choice in Africa: Are Institutional Infrastructure and Non-Traditional Determinants Important? International Review of Financial Analysis, 31:152–166.
- Hadlock, C. J. and Pierce, J. R. (2010). New Evidence on Measuring Financial Constraints: Moving Beyond the KZ Index. *The Review of Financial Studies*, 23(5):1909– 1940.
- He, W., Ng, L., Zaiats, N., and Zhang, B. (2017). Dividend policy and earnings management across countries. *Journal of Corporate Finance*, 42(C):267–286. Publisher: Elsevier.
- Hovakimian, A., Hovakimian, G., and Tehranian, H. (2004). Determinants of Target Capital Structure: The Case of Dual Debt and Equity Issues. *Journal of Financial Economics*, 71(3):517–540.
- Hu, X. and Schiantarelli, F. (1998). Investment and Capital Market Imperfections: A Switching Regression Approach Using U.S. Firm Panel Data. *Review of Economics* and Statistics, 80(3):466–479.
- Iturriaga, F. J. L. and Crisóstomo, V. L. (2010). Do Leverage, Dividend Payout, and Ownership Concentration Influence Firms' Value Creation? An Analysis of Brazilian Firms. *Emerging Markets Finance and Trade*, 46(3):80–94. Publisher: Routledge _eprint: https://doi.org/10.2753/REE1540-496X460306.
- Kaplan, S. N. and Zingales, L. (1997). Do Investment-Cash Flow Sensitivities Provide Useful Measures of Financing Constraints? The Quarterly Journal of Economics, 112(1):169–215.
- Khurana, I. K., Martin, X., and Pereira, R. (2006). Financial Development and the Cash

Flow Sensitivity of Cash. Journal of Financial and Quantitative Analysis, 41(04):787–808.

- Khémiri, W. and Noubbigh, H. (2019). Does sub-Saharan Africa overinvest? Evidence from a panel of non-financial firms. *The Quarterly Review of Economics and Finance*.
- Levine, R. (1997). Financial Development and Economic Growth: Views and Agenda. Journal of Economic Literature, 35(2):688–726. Publisher: American Economic Association.
- Lewellen, J. and Lewellen, K. (2016). Investment and Cash Flow: New Evidence. *Journal* of Financial and Quantitative Analysis, 51(4):1135–1164.
- Machokoto, M. and Areneke, G. (2020). Is the cash flow sensitivity of cash asymmetric? African evidence. *Finance Research Letters*, page 101440.
- Machokoto, M., Tanveer, U., Ishaq, S., and Areneke, G. (2019). Decreasing investmentcash flow sensitivity: Further UK evidence. *Finance Research Letters*, page 101397.
- McLean, R. D. and Zhao, M. (2018). Cash savings and capital markets. Journal of Empirical Finance, 47:49–64.
- Moshirian, F., Nanda, V., Vadilyev, A., and Zhang, B. (2017). What drives investment–cash flow sensitivity around the World? An asset tangibility Perspective. *Journal* of Banking & Finance, 77:1–17.
- Moyen, N. (2007). How Big is the Debt Overhang Problem? Journal of Economic Dynamics and Control, 31(2):433–472.
- Mu, Y., Phelps, P., and Stotsky, J. G. (2013). Bond markets in Africa. Review of Development Finance, 3(3):121–135.

- Oztekin, O. (2015). Capital Structure Decisions around the World: Which Factors Are Reliably Important? *Journal of Financial and Quantitative Analysis*, 50(03):301–323.
- Panibratov, A. (2017). International Strategy of Emerging Market Firms : Absorbing Global Knowledge and Building Competitive Advantage. Routledge, London, UK, 1 edition.
- Popov, A. and Rocholl, J. (2018). Do credit shocks affect labor demand? Evidence for employment and wages during the financial crisis. *Journal of Financial Intermediation*, 36:16–27.
- Riddick, L. A. and Whited, T. M. (2009). The Corporate Propensity to Save. The Journal of Finance, 64(4):1729–1766.
- Roodman, D. (2009). A Note on the Theme of Too Many Instruments. Oxford Bulletin of Economics and Statistics, 71(1):135–158.
- Shirai, S. (2004). Testing the Three Roles of Equity Markets in Developing Countries: The Case of China. World Development, 32(9):1467–1486.
- Sorge, M., Zhang, C., and Koufopoulos, K. (2017). Short-Term Corporate Debt around the World. Journal of Money, Credit and Banking, 49(5):997–1029.
- Wesson, N., Bruwer, B. W., and Hamman, W. D. (2015). Share repurchase and dividend payout behaviour: The South African experience. South African Journal of Business Management, 46(3):43–54. Number: 3.
- Whited, T. M. and Wu, G. (2006). Financial Constraints Risk. The Review of Financial Studies, 19(2):531–559.



Figure 1 Financial development and economic growth across countries

The figure presents a scatter plot of stock market capitalisation-to-GDP (%) and private credit-to-GDP (%) with superimposed average GDP (constant 2010 USD). The depicted emerging and developed countries (FIC Codes) are Australia (AUS), Brazil (BRA), Canada (CAN), China (CHN), Egypt (EGY), France (FRA), Germany (DEU), India (IND), Ivory Coast (CIV), Japan (JPN), Kenya (KEN), Morocco (MAR), Nigeria (NGA), Singapore (SGP), South Africa (ZAF), Switzerland (CHE), Tunisia (TUN), UK (GBR) and USA (USA). The data is drawn from *The World Bank* over the period 2000–2015. All variables used are defined in Appendix A.

Table 1 Basic statistics

The table presents the summary statistics of the variables used. The sample consists of listed non-utility and non-financial firms in selected African countries drawn from *Datastream* over the period 2000–2015. All variables used are defined in Appendix A and are winsorised at the lower and upper one percentiles. ***, **, * indicate significance at the one, five, and ten percent levels, respectively.

| # | Variables | Ν | Mean | Std.Dev | Min | p25 | Median | p75 | Max | Trend |
|------|---------------------|-----------|---------|---------|---------|---------|---------|---------|--------|----------------|
| (1) | $\Delta Cash$ | $5,\!940$ | 0.018 | 0.079 | -0.394 | -0.019 | 0.008 | 0.047 | 0.679 | -0.021 |
| (2) | Capex | 5,940 | 0.079 | 0.063 | 0.000 | 0.035 | 0.064 | 0.107 | 0.529 | -0.156^{***} |
| (3) | Div | 5,940 | 0.054 | 0.067 | 0.000 | 0.013 | 0.034 | 0.070 | 0.730 | 0.085^{**} |
| (4) | $\Delta \mathrm{D}$ | 5,940 | 0.019 | 0.084 | -0.525 | -0.016 | 0.000 | 0.045 | 0.802 | 0.145^{***} |
| (5) | ΔE | 5,940 | -0.061 | 0.110 | -0.795 | -0.102 | -0.057 | -0.020 | 0.732 | 0.157^{***} |
| (6) | CF | 5,940 | 0.156 | 0.100 | 0.000 | 0.085 | 0.138 | 0.208 | 0.703 | -0.299*** |
| (7) | q | $5,\!940$ | 1.727 | 1.155 | 0.284 | 1.090 | 1.459 | 2.060 | 50.535 | 3.420^{***} |
| (8) | \overline{SG} | $5,\!940$ | 0.126 | 0.188 | -0.496 | 0.034 | 0.110 | 0.203 | 0.976 | -0.832*** |
| (9) | Size | $5,\!940$ | 15.255 | 1.974 | 8.039 | 13.817 | 15.455 | 16.671 | 19.294 | 4.925^{***} |
| (10) | PPE | 5,940 | 0.364 | 0.222 | 0.009 | 0.171 | 0.331 | 0.546 | 0.977 | -0.009 |
| (11) | WW | 5,940 | -0.734 | 0.096 | -0.991 | -0.801 | -0.748 | -0.666 | -0.370 | -0.227*** |
| (12) | HP | 5,940 | -11.497 | 0.943 | -12.369 | -12.135 | -12.079 | -10.989 | -6.765 | -3.283*** |
| (13) | LogAge | $5,\!940$ | 2.436 | 0.516 | 1.099 | 2.079 | 2.485 | 2.833 | 3.258 | 8.511*** |

Panel A: Main variables

| 1 un | er D i Dinere | | bub bump | leb | | | | |
|---------------|----------------------|----------------|------------------------------------------------|-----------------------------------------------|-----------------------------------------------|-----------------------------------------------|-----------------------------------------------|----------------------|
| | | Variables | \mathbf{CF} | $\Delta Cash$ | Capex | Div | ΔDebt | $\Delta Equity$ |
| \mathbf{FC} | Category | Metric | (1) | (2) | (3) | (4) | (5) | (6) |
| | Low | Mean Median | $0.156 \\ 0.137$ | $0.012 \\ 0.005$ | $0.086 \\ 0.074$ | $0.062 \\ 0.038$ | $0.025 \\ 0.004$ | -0.068 -0.061 |
| ΜM | High | Mean Median | $0.157 \\ 0.141$ | $0.025 \\ 0.012$ | $\begin{array}{c} 0.072 \\ 0.056 \end{array}$ | $0.047 \\ 0.029$ | $0.012 \\ 0.000$ | -0.054 -0.053 |
| | Diff <i>p</i> -value | Mean Median | $\begin{bmatrix} 0.767 \\ 0.027 \end{bmatrix}$ | [0.000] [0.000] | [0.000] [0.000] | [0.000] [0.000] | [0.000] [0.000] | [0.000] [0.000] |
| | Low | Mean Median | $\begin{array}{c} 0.152 \\ 0.134 \end{array}$ | $0.013 \\ 0.005$ | $0.086 \\ 0.074$ | $\begin{array}{c} 0.058\\ 0.035\end{array}$ | $0.027 \\ 0.006$ | -0.062 -0.059 |
| HP | High | Mean Median | $\begin{array}{c} 0.162 \\ 0.147 \end{array}$ | $\begin{array}{c} 0.024 \\ 0.012 \end{array}$ | $\begin{array}{c} 0.070 \\ 0.053 \end{array}$ | $\begin{array}{c} 0.050\\ 0.031 \end{array}$ | $0.008 \\ 0.000$ | -0.060 -0.054 |
| | Diff <i>p</i> -value | Mean Median | [0.000] [0.000] | [0.000] [0.000] | [0.000] [0.000] | [0.000] [0.001] | [0.000] [0.000] | [0.525] [0.003] |
| | Low | Mean Median | $\begin{array}{c} 0.169 \\ 0.153 \end{array}$ | $\begin{array}{c} 0.022 \\ 0.011 \end{array}$ | $\begin{array}{c} 0.073 \\ 0.056 \end{array}$ | $\begin{array}{c} 0.058\\ 0.035\end{array}$ | $\begin{array}{c} 0.010\\ 0.000 \end{array}$ | -0.066 -0.058 |
| Size | High | Mean Median | $0.143 \\ 0.125$ | $0.013 \\ 0.005$ | $0.086 \\ 0.074$ | $\begin{array}{c} 0.050\\ 0.031 \end{array}$ | $0.028 \\ 0.008$ | -0.056 -0.057 |
| | Diff <i>p</i> -value | Mean Median | $[0.000] \\ [0.000]$ | [0.000] [0.000] | [0.000] [0.000] | [0.000] [0.009] | [0.000] [0.000] | [0.000] [0.468] |
| | Low | Mean Median | $0.165 \\ 0.147$ | $0.023 \\ 0.010$ | $\begin{array}{c} 0.076 \\ 0.059 \end{array}$ | $0.053 \\ 0.033$ | $\begin{array}{c} 0.015 \\ 0.000 \end{array}$ | -0.059 -0.056 |
| ogAge | High | Mean Median | $0.145 \\ 0.128$ | $0.012 \\ 0.005$ | $0.083 \\ 0.071$ | $\begin{array}{c} 0.056 \\ 0.035 \end{array}$ | $0.024 \\ 0.007$ | -0.064 -0.059 |
| Г | Diff <i>p</i> -value | Mean Median | $[0.000] \\ [0.000]$ | [0.000] [0.000] | [0.000] [0.000] | $[0.104] \\ [0.095]$ | [0.000] [0.000] | [0.129] [0.002] |
| | Pre-Crisis | Mean Median | $0.168 \\ 0.147$ | $0.024 \\ 0.010$ | $0.084 \\ 0.065$ | $\begin{array}{c} 0.054 \\ 0.034 \end{array}$ | $\begin{array}{c} 0.018\\ 0.000\end{array}$ | -0.071 -0.067 |
| Crisis | Crisis | Mean Median | $0.155 \\ 0.139$ | $0.013 \\ 0.005$ | $0.079 \\ 0.064$ | $0.059 \\ 0.038$ | $0.013 \\ 0.000$ | -0.062 -0.058 |
| C | Diff <i>p</i> -value | Mean Median | [0.000] [0.005] | [0.000] [0.001] | [0.009] [0.533] | [0.016] [0.012] | [0.045] [0.652] | $[0.010] \\ [0.000]$ |
| s | Others | Mean Median | $0.174 \\ 0.154$ | 0.012 0.005 | $\begin{array}{c} 0.079 \\ 0.057 \end{array}$ | $0.083 \\ 0.063$ | 0.011 0.000 | -0.073 -0.067 |
| untrie | S.Africa | Mean Median | $0.152 \\ 0.136$ | $0.019 \\ 0.008$ | $0.079 \\ 0.065$ | $0.048 \\ 0.030$ | $0.021 \\ 0.001$ | -0.058 -0.056 |
| Ğ | Diff <i>p</i> -value | Mean Median | $[0.000] \\ [0.000]$ | [0.003] [0.100] | [0.975] [0.023] | [0.000] [0.000] | $[0.001] \\ [0.000]$ | $[0.000] \\ [0.000]$ |

 Table 1 Basic statistics (continued)

| The tε non-fir A and | vble presents nancial firms are winsoris | the pairwise 5 in selected Afi ed at the lower | Spearman (Pear rican countries d and upper one] | son) correlation lrawn from <i>Dat</i> percentiles. *** | s in the above astream over th , **, * indicate | (below) diagona e period 2000–2 significance at th | The sample 015. All variable ne one, five, and | consists of listed les used are defi ten percent lev | d non-utility and ned in Appendix els, respectively. |
|----------------------------|------------------------------------------------|------------------------------------------------------|--------------------------------------------------------|---------------------------------------------------------------|-------------------------------------------------------|----------------------------------------------------------|----------------------------------------------------------------------------|------------------------------------------------------------|------------------------------------------------------------|
| # | Variables | (1) | (2) | (3) | (4) | (5) | (9) | (2) | (8) |
| (1) | $\Delta \mathrm{Cash}$ | 1 | -0.103^{***} | -0.037*** | 0.019 | 0.160^{***} | 0.314^{***} | -0.017 | 0.068^{***} |
| (2) | Capex | -0.093*** | 1 | 0.049^{***} | 0.235^{***} | -0.026^{**} | 0.285^{***} | 0.199^{***} | 0.088^{***} |
| (3) | Div | -0.052^{***} | 0.079^{***} | 1 | 0.055^{***} | -0.526^{***} | 0.469^{***} | 0.581^{***} | 0.059^{***} |
| (4) | ΔD | 0.080^{***} | 0.262^{***} | 0.067^{***} | 1 | 0.030^{**} | -0.110^{***} | 0.146^{***} | 0.088^{***} |
| $(\overline{5})$ | $\Delta \mathrm{E}$ | 0.199^{***} | -0.036^{***} | -0.522^{***} | -0.043^{***} | 1 | -0.306^{***} | -0.307^{***} | -0.044^{***} |
| (0) | CF | 0.319^{***} | 0.300^{***} | 0.573^{***} | -0.036^{***} | -0.290^{***} | 1 | 0.429^{***} | 0.175^{***} |
| (-1) | q | 0.033^{**} | 0.143^{***} | 0.472^{***} | 0.120^{***} | -0.187^{***} | 0.381^{***} | -1 | 0.167^{***} |
| (8) | SG | 0.071^{***} | 0.117^{***} | 0.083^{***} | 0.075^{***} | -0.030^{**} | 0.197^{***} | 0.087^{***} | 1 |
| (6) | \mathbf{Size} | -0.096*** | 0.127^{***} | -0.004 | 0.076^{***} | 0.013 | -0.071^{***} | 0.063^{***} | -0.031^{**} |
| (10) | PPE | -0.049^{***} | 0.393^{***} | -0.041^{***} | -0.014 | 0.026^{**} | 0.060^{***} | -0.025^{*} | -0.061^{***} |
| (11) | WM | 0.087^{***} | -0.134^{***} | -0.116^{***} | -0.083*** | 0.039^{***} | -0.008 | -0.125^{***} | 0.042^{***} |
| (12) | HP | 0.054^{***} | -0.096^{***} | -0.049^{***} | -0.107^{***} | 0.014 | 0.047^{***} | -0.079*** | 0.021 |
| (13) | LogAge | -0.043*** | -0.008 | -0.048*** | 0.076^{***} | 0.062^{***} | -0.158*** | 0.027^{**} | -0.150^{***} |
| | | | | | | | | | |
| # | Variables | (6) | (10) | (11) | (12) | (13) | | | |
| (1) | $\Delta \mathrm{Cash}$ | -0.108^{***} | -0.047^{***} | 0.102^{***} | 0.061^{***} | -0.048^{***} | 1 | | |
| $(\overline{2})$ | Capex | 0.200^{***} | 0.448^{***} | -0.199^{***} | -0.188^{***} | 0.059^{***} | | | |
| (3) | Div | -0.014 | -0.097*** | -0.142^{***} | -0.052^{***} | -0.039^{***} | | | |
| (4) | $\Delta \mathrm{D}$ | 0.099^{***} | 0.000 | -0.097*** | -0.139^{***} | 0.105^{***} | | | |
| (2) | $\Delta \mathrm{E}$ | 0.051^{***} | 0.083^{***} | 0.026^{**} | -0.019 | 0.097^{***} | | | |
| (9) | CF | -0.094*** | 0.055^{***} | 0.009 | 0.062^{***} | -0.159^{***} | | | |
| (-) | d | 0.129^{***} | -0.081^{***} | -0.201^{***} | -0.167^{***} | 0.107^{***} | | | |
| (8) | SG | -0.034^{***} | -0.084*** | 0.032^{**} | 0.051^{***} | -0.134*** | | | |
| (6) | Size | 1 | 0.278^{***} | -0.950*** | -0.850*** | 0.445^{***} | | | |
| (10) | PPE | 0.253^{***} | 1 | -0.255*** | -0.184^{***} | 0.089^{***} | | | |
| (11) | | -0.944*** 0.000*** | -0.240*** 0.125*** | Т О о <i>г</i> ожжж | 0.821^{***} | -0.428*** | | | |
| (13) | пг LogAge | 0.425^{***} | 0.069*** | 0.002 -0.410*** | -0.456^{***} | -0.402 1 | | | |
| | | | | | | | 1 | | |

 Table 2 Correlations

| 1 |
|---------------------|
| . <u> </u> |
| E |
| ·= |
| > |
| •– |
| ÷ |
| • • |
| <u>o</u> |
| F |
| Ð |
| ō |
| |
| ⊳ |
| 2 |
| 0 |
| E |
| - |
| _ |
| |
| 4 |
| lsh |
| ash |
| Cash |
| Cash |
| Cash |
| 3 Cash |
| 3 Cash |
| le 3 Cash |
| ole 3 Cash |
| ble 3 Cash |
| able 3 Cash |
| Table 3 Cash |

The table presents the estimation results of a system of equations depicted by Equation (1) relating the uses-of-funds to cash flow and firm characteristics. Panel A presents the main models. Panel B presents the Spearman (above diagonal) and Pearson (below diagonal) pairwise correlations of the residuals across the equations. The sample consists of listed non-utility and non-financial firms in selected African countries drawn from *Datastream* over the period 2000–2015. All variables used are defined in Appendix A and are winsorised at the lower and upper one percentiles. ***, **, * indicate significance at the one, five, and ten percent levels, respectively. ρ

| models |
|--------|
| Main 1 |
| Y: |
| anel |

| Fanel A: N | lain models | | | | | | | | | |
|--------------------------------|----------------|----------------|----------------|----------------|----------------|----------------|----------------|----------------|----------------|----------------|
| | | Model | s without con | straints | | | Mode | ls with consti | raints | |
| | $\Delta Cash$ | Capex | Div | ΔD | ΔE | $\Delta Cash$ | Capex | Div | ΔD | ΔE |
| Variables | (1) | (2) | (3) | (4) | (5) | (9) | (2) | (8) | (6) | (10) |
| CF_{ijt} | 0.438^{***} | 0.084*** | 0.177^{***} | -0.136*** | -0.102*** | 0.444^{***} | 0.084*** | 0.185^{***} | -0.152^{***} | -0.135^{***} |
| 5 | (0.013) | (0.007) | (0.007*** | (0.014) | (0.017) | (0.013) | (0.007) | (0.006) | (0.013) | (0.012) |
| q_{ijt-1} | (0.001) | (0.001) | (0.001) | (0.001) | (0.001) | (0.001) | (0.001) | (0.001) | (0.001) | (0.001) |
| SG_{iit-1} | -0.002 | 0.001 | 0.020^{***} | 0.001 | -0.016^{**} | -0.005 | 0.001 | 0.015^{***} | 0.009* | 0.002 |
| <i>و</i> | (0.005) | (0.003) | (0.003) | (0.006) | (0.007) | (0.005) | (0.003) | (0.003) | (0.006) | (0.005) |
| ${ m Size}_{ijt}$ | -0.014^{***} | -0.013^{***} | 0.012^{***} | -0.034^{***} | -0.029^{***} | -0.018^{***} | -0.013^{***} | 0.006^{***} | -0.022^{***} | -0.004 |
| 5 | (0.003) | (0.002) | (0.002) | (0.004) | (0.005) | (0.003) | (0.002) | (0.002) | (0.004) | (0.003) |
| PPE_{ijt} | 0.149^{***} | -0.096*** | -0.115^{***} | -0.025 | 0.142^{***} | 0.166^{***} | -0.096*** | -0.092^{***} | -0.070*** | 0.048^{***} |
| 5 | (0.016) | (0.00) | (0.000) | (0.018) | (0.021) | (0.016) | (0.009) | (0.008) | (0.016) | (0.015) |
| $\Delta \mathrm{Cash}_{ijt-1}$ | -0.228^{***} | 0.001 | 0.025^{***} | 0.023^{*} | -0.065^{***} | -0.212^{***} | 0.001 | 0.046^{***} | -0.017 | -0.149^{***} |
| 2 | (0.012) | (0.007) | (0.006) | (0.013) | (0.015) | (0.011) | (0.007) | (0.006) | (0.012) | (0.011) |
| $\operatorname{Capex}_{iit-1}$ | -0.113^{***} | 0.370^{***} | 0.029^{**} | 0.272^{***} | -0.075^{***} | -0.122^{***} | 0.370^{***} | 0.018 | 0.294^{***} | -0.028 |
| , ç | (0.022) | (0.013) | (0.012) | (0.025) | (0.029) | (0.022) | (0.013) | (0.011) | (0.023) | (0.021) |
| ${ m Div}_{ijt-1}$ | -0.136^{***} | 0.019 | 0.239^{***} | 0.207^{***} | -0.204^{***} | -0.148^{***} | 0.019 | 0.224^{***} | 0.237^{***} | -0.142^{***} |
| 5 | (0.023) | (0.014) | (0.013) | (0.026) | (0.031) | (0.023) | (0.014) | (0.012) | (0.024) | (0.023) |
| $\Delta \mathrm{D}_{ijt-1}$ | 0.024^{**} | -0.017^{**} | -0.080*** | -0.098*** | 0.087^{***} | 0.030^{**} | -0.017^{**} | -0.072^{***} | -0.114^{***} | 0.055^{***} |
| 5 | (0.012) | (0.007) | (0.007) | (0.014) | (0.016) | (0.012) | (0.007) | (0.006) | (0.012) | (0.012) |
| $\Delta \mathrm{E}_{ijt-1}$ | 0.064^{***} | -0.035^{***} | -0.023*** | 0.008 | 0.035^{**} | 0.067^{***} | -0.035^{***} | -0.018^{***} | -0.001 | 0.015 |
| 3 | (0.011) | (0.006) | (0.006) | (0.012) | (0.014) | (0.011) | (0.006) | (0.005) | (0.011) | (0.010) |
| N | 5.940 | 5.940 | 5.940 | 5,940 | 5.940 | 5.940 | 5.940 | 5.940 | 5.940 | 5.940 |
| R^{2} | 0.24 | 0.19 | 0.29 | 0.06 | 0.04 | 0.24 | 0.19 | 0.28 | 0.06 | 0.03 |

| sh | $\Delta Cash$ 1.000*** -0.184*** | Models Capex -0.217*** 1.000*** | s without con Div -0.234*** | $\begin{array}{c} \text{straints} \\ \Delta D \\ \Delta D \\ 0.164^{***} \\ 0.313^{***} \end{array}$ | ΔE 0.240*** 0.026** | $\Delta Cash$ 1.000*** -0.184*** | Mode Capex -0.216*** | ls with const Div -0.227*** | raints ΔD 0.165^{***} | ΔE 0.237*** 0.017**** |
|----|----------------------------------------|------------------------------------------|-----------------------------------|------------------------------------------------------------------------------------------------------|-----------------------------------|----------------------------------------|----------------------------|-----------------------------------|------------------------------------|-----------------------------|
| | -0.222^{***} | 0.003 | 1.000^{***} | 0.106^{**} | -0.343^{+++} | -0.219^{***} | 0.003 | 1.000^{***} | 0.102^{++} | -0.354^{***} |
| | 0.145^{***} | 0.296^{***} | 0.117^{***} | 1.000^{***} | -0.007 | 0.143^{***} | 0.295^{***} | 0.111*** | 1.000^{++} | -0.001 |
| | 0.253^{***} | 0.057^{***} | -0.371^{***} | - 0.069^{***} | 1.000^{+++} | 0.246^{***} | 0.056^{***} | -0.377*** | -0.058^{++} | 1.000^{***} |

Table 3 Cash flow sensitivities

Table 4 The effect of credit constraints on investment and financing-cash flow sensitivities

The table presents the estimation results of a system of equations depicted by Equation (1) relating the uses-of-funds to cash flow and firm characteristics. All models include control variables and the lagged use-of-funds (but not reported). The correlation of residuals across the system of equations is not reported for brevity. The sample consists of listed non-utility and non-financial firms in selected African countries drawn from *Datastream* over the period 2000–2015. All variables used are defined in Appendix A and are winsorised at the lower and upper one percentiles. ***, **, * indicate significance at the one, five, and ten percent levels, respectively.

| | | $\Delta Cash$ | Capex | Div | ΔD | ΔE |
|---------|---------------------------|---------------------------------------------------------------------------------|---------------------------------------------------------------------------------|---------------------------------------------------------------------------------|-----------------------------------------------------------------------------------|-----------------------------------------------------------------------------------|
| FC | Variables | (1) | (2) | (3) | (4) | (5) |
| W Index | CF^{Low} CF^{High} | $\begin{array}{c} 0.351^{***} \\ (0.013) \\ 0.540^{***} \\ (0.012) \end{array}$ | $\begin{array}{c} 0.106^{***} \\ (0.007) \\ 0.064^{***} \\ (0.007) \end{array}$ | $\begin{array}{c} 0.254^{***} \\ (0.007) \\ 0.099^{***} \\ (0.006) \end{array}$ | $\begin{array}{c} -0.135^{***} \\ (0.014) \\ -0.191^{***} \\ (0.012) \end{array}$ | $\begin{array}{c} -0.154^{***} \\ (0.013) \\ -0.106^{***} \\ (0.012) \end{array}$ |
| Μ | Diff p -value | [0.000] | [0.000] | [0.000] | [0.003] | [0.006] |
| P Index | CF^{Low} CF^{High} | $\begin{array}{c} 0.358^{***} \\ (0.013) \\ 0.554^{***} \\ (0.012) \end{array}$ | $\begin{array}{c} 0.115^{***} \\ (0.007) \\ 0.047^{***} \\ (0.007) \end{array}$ | $\begin{array}{c} 0.241^{***} \\ (0.006) \\ 0.090^{***} \\ (0.006) \end{array}$ | $\begin{array}{c} -0.172^{***} \\ (0.015) \\ -0.156^{***} \\ (0.012) \end{array}$ | -0.115*** (0.013) -0.153*** (0.012) |
| Η | Diff <i>p</i> -value | [0.000] | [0.000] | [0.000] | [0.402] | [0.026] |
| Size | CF^{Low} CF^{High} | $\begin{array}{c} 0.514^{***} \\ (0.012) \\ 0.348^{***} \\ (0.013) \end{array}$ | $\begin{array}{c} 0.053^{***} \\ (0.007) \\ 0.122^{***} \\ (0.008) \end{array}$ | $\begin{array}{c} 0.145^{***} \\ (0.006) \\ 0.226^{***} \\ (0.006) \end{array}$ | $\begin{array}{c} -0.134^{***} \\ (0.012) \\ -0.188^{***} \\ (0.015) \end{array}$ | $\begin{array}{c} -0.155^{***} \\ (0.012) \\ -0.115^{***} \\ (0.013) \end{array}$ |
| | Diff <i>p</i> -value | [0.000] | [0.000] | [0.000] | [0.004] | [0.024] |
| LogAge | CF^{Low} CF^{High} | $\begin{array}{c} 0.533^{***} \\ (0.013) \\ 0.340^{***} \\ (0.012) \end{array}$ | $\begin{array}{c} 0.061^{***} \\ (0.007) \\ 0.124^{***} \\ (0.007) \end{array}$ | $\begin{array}{c} 0.108^{***} \\ (0.006) \\ 0.268^{***} \\ (0.007) \end{array}$ | -0.212*** (0.013) -0.068*** (0.014) | -0.085*** (0.012) -0.200*** (0.012) |
| | Diff p -value | [0.000] | [0.000] | [0.000] | [0.000] | [0.000] |

| crisis |
|---------------|
| ancial |
| fin |
| the |
| around |
| sensitivities |
| flow |
| Cash |
| ນ |
| Table |

The table presents the estimation results of a system of equations depicted by Equation (1) relating the uses-of-funds to cash flow and firm characteristics. Panel A presents the main models. Panel B presents the Spearman (above diagonal) and Pearson (below diagonal) pairwise correlations of the residuals across the equations. All models include control variables and the lagged use-of-funds (but not reported). The correlation of residuals across the system of equations is not reported for brevity. The sample consists of listed non-utility and non-financial firms in selected African countries drawn from *Datastream* over the period 2000–2015. All variables used are defined in Appendix A and are winsorised at the lower and upper one percentiles. ***, **, * indicate significance at the one, five, and ten percent levels, respectively.

| | | FC Variable | AII CF AII CF Pre vs (| $\begin{array}{c} \operatorname{CF}^{Low} \\ \operatorname{Pre} vs \end{array} ($ | $WW \stackrel{\text{LF}}{=} CF^{High}$ $WW \stackrel{\text{Pre } vs}{=} ($ | Low vs. | CF ^{Low} ex Pre vs (| $\begin{array}{c} \operatorname{HP} \operatorname{Ind} & \\ \operatorname{CF}^{High} \\ \end{array} \\ \operatorname{Pre} vs \operatorname{C} \end{array}$ |
|-----------------|---------------|-------------------|-------------------------------------------------------------------------|-----------------------------------------------------------------------------------|----------------------------------------------------------------------------|-------------|------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------|
| | ΔCas | (1) | (1) 0.446 (0.010 risis | 0.324 (0.02) risis | 0.584 (0.01: risis | ligh [0.000 | $\begin{array}{c} 0.328 \\ (0.02 \end{array} \end{array}$ | 0.559 (0.02) risis |
| 1 | sh | sh | 6) 6) | [*** [5] | [*** 9) | 0] | }*** 4) |)*** (0 |
| Bef | Capex | Capex | (2) 0.069*** (0.008) | 0.052^{***} (0.011) | 0.067^{**} (0.012) | [0.173] | 0.090^{**} (0.011) | 0.047^{***} (0.012) |
| ore (2002–20 | Div | Div (3) | (3) 0.142^{***} (0.008) | 0.197^{***} (0.012) | 0.045^{***} (0.011) | [0.000] | 0.211^{***} (0.011) | 0.068^{**} (0.012) |
| (200 | ΔD | (4) | (4) -0.161*** (0.016) | -0.243^{***} (0.023) | -0.157^{***} (0.021) | [0.688] | -0.188^{***} (0.023) | -0.167^{***} (0.021) |
|) | ΔE | ΔE (5) | (5) -0.182*** (0.014) | -0.184^{***} (0.021) | -0.147^{***} (0.018) | [0.392] | -0.183^{***} (0.020) | -0.159^{***} (0.019) |
| | $\Delta Cash$ | $\Delta Cash$ (6) | $\begin{array}{c} (0) \\ 0.467^{***} \\ (0.014) \\ [0.316] \end{array}$ | $\begin{array}{c} 0.419^{***} \\ (0.019) \\ [0.002] \end{array}$ | $\begin{array}{c} 0.532^{***} \\ (0.020) \\ [0.066] \end{array}$ | [0.000] | $\begin{array}{c} 0.403^{***} \\ (0.017) \\ [0.011] \end{array}$ | $\begin{array}{c} 0.578^{***} \\ (0.024) \\ [0.545] \end{array}$ |
| Af | Capex | Capex (7) | (1) 0.101*** (0.009) [0.009] | $\begin{array}{c} 0.129^{***} \\ (0.013) \\ [0.000] \end{array}$ | $\begin{array}{c} 0.080^{***} \\ (0.013) \\ [0.427] \end{array}$ | [0.000] | $\begin{array}{c} 0.142^{***} \\ (0.012) \\ [0.002] \end{array}$ | $\begin{array}{c} 0.043^{***} \\ (0.014) \\ [0.834] \end{array}$ |
| ter $(2008-20)$ | Div | Div (8) | (δ) 0.179*** (0.007) [0.001] | $\begin{array}{c} 0.240^{***} \\ (0.010) \\ [0.007] \end{array}$ | $\begin{array}{c} 0.112^{***} \\ (0.009) \\ [0.000] \end{array}$ | [0.000] | $\begin{array}{c} 0.212^{***} \\ (0.009) \\ [0.959] \end{array}$ | $\begin{array}{c} 0.115^{***} \\ (0.011) \\ [0.003] \end{array}$ |
| 12) | ΔD | (10) (6) | (9) -0.153*** (0.015) [0.719] | -0.116^{***} (0.023) [0.000] | -0.211^{***} (0.020) [0.059] | [0.012] | -0.156^{***} (0.021) [0.313] | -0.179^{***} (0.021) [0.685] |
| , | ΔE | QE (10) | $\begin{array}{c} (10) \\ -0.100 \\ (0.015 \\ 0.000 \end{array}$ | -0.096 (0.020) [0.003] | -0.065° (0.021) [0.003] | [0.826] | -0.087* (0.018) (0.018) $[0.000]$ | -0.085° (0.025) [0.017] |

| | | | Bef | ore (2002–20 | (200 | | | Afi | ter (2008–20 | 12) | |
|-------------------|------------------------------------|--------------------------|--------------------------|-------------------------------------------------------|------------------------|------------------------|------------------------------------------------------------------|------------------------------------------------------------------|------------------------------------------------------------------|-----------------------------------------------------------|--------------------------------------|
| | | $\Delta Cash$ | Capex | Div | ΔD | ΔE | $\Delta Cash$ | Capex | Div | ΔD | ΔE |
| FC | Variables | (1) | (2) | (3) | (4) | (5) | (9) | (7) | (8) | (6) | (10) |
| ; | CF ^{Low} Pre vs Crisis | 0.548^{***} (0.019) | 0.056^{***} (0.011) | 0.094^{***} (0.011) | -0.125^{***} (0.020) | -0.177^{***} (0.017) | $\begin{array}{c} 0.508^{***} \\ (0.021) \\ [0.154] \end{array}$ | $\begin{array}{c} 0.061^{***} \\ (0.012) \\ [0.755] \end{array}$ | $\begin{array}{c} 0.188^{***} \\ (0.010) \\ [0.000] \end{array}$ | -0.112^{***} (0.019) [0.648] | -0.130^{***} (0.021) [0.086] |
| \overline{sziS} | CF^{High} Pre vs Crisis | 0.285^{***} (0.027) | 0.095^{***} (0.012) | 0.199^{***} (0.012) | -0.255^{***} (0.025) | -0.166^{***} (0.023) | $\begin{array}{c} 0.417^{***} \\ (0.018) \\ [0.000] \end{array}$ | $\begin{array}{c} 0.135^{***} \\ (0.015) \\ [0.034] \end{array}$ | $\begin{array}{c} 0.153^{***} \\ (0.009) \\ [0.002] \end{array}$ | -0.253^{***} (0.023) [0.963] | -0.041^{**} (0.021) [0.000] |
| | Low vs High | [0.000] | [0.210] | [0.000] | [0.972] | [0.372] | [0.000] | [0.000] | [0.000] | [0.002] | [0.144] |
| ge | CF ^{Low} Pre vs Crisis | 0.541^{***} (0.021) | 0.071^{***} (0.011) | 0.083^{**} (0.010) | -0.197^{***} (0.020) | -0.109^{***} (0.017) | $\begin{array}{c} 0.535^{***} \\ (0.019) \\ [0.839] \end{array}$ | $\begin{array}{c} 0.073^{***} \\ (0.012) \\ [0.916] \end{array}$ | $\begin{array}{c} 0.104^{***} \\ (0.008) \\ [0.110] \end{array}$ | -0.220^{***} (0.018) [0.393] | -0.069^{***} (0.020) [0.128] |
| AgoJ | CF^{High} Pre vs Crisis | 0.357^{***} (0.023) | 0.074^{***} (0.012) | $\begin{array}{c} 0.207^{***} \\ (0.013) \end{array}$ | -0.133^{***} (0.026) | -0.229^{***} (0.023) | $\begin{array}{c} 0.384^{***} \\ (0.020) \\ [0.379] \end{array}$ | $\begin{array}{c} 0.149^{***} \\ (0.014) \\ [0.000] \end{array}$ | $\begin{array}{c} 0.300^{***} \\ (0.011) \\ [0.000] \end{array}$ | $\begin{array}{c} -0.018\\ (0.025)\\ [0.001] \end{array}$ | -0.150^{***} (0.022) [0.013] |
| | Low vs High | [0.001] | [0.000] | [0.000] | [0.059] | [0.136] | [0.000] | [0.000] | [0000] | [0.00] | [0.002] |

| \sim |
|-----------------------|
| (continued) |
| crisis |
| financial |
| \mathbf{the} |
| around |
| sensitivities |
| flow |
| Cash |
| Ŋ |
| able |

Table 6 The effects of deviating from target and non-linearities in cash flow sensitivities

The table presents the estimation results of a system of equations depicted by Equation (1) relating the uses-of-funds to cash flow and firm characteristics. All models include control variables and the lagged use-of-funds (but not reported). The correlation of residuals across the system of equations is not reported for brevity. The sample consists of listed non-utility and non-financial firms in selected African countries drawn from *Datastream* over the period 2000–2015. All variables used are defined in Appendix A and are winsorised at the lower and upper one percentiles. ***, **, * indicate significance at the one, five, and ten percent levels, respectively.

| | | $\Delta Cash$ | Capex | Div | ΔD | ΔE |
|----------|--------------------------------------------|---------------------------------------------------------------------------------|---------------------------------------------------------------------------------|---------------------------------------------------------------------------------|-----------------------------------------------------------------------------------|-----------------------------------------------------------------------------------|
| Proxy | Variables | (1) | (2) | (3) | (4) | (5) |
| Cash | CF^{Below} CF^{Above} | $\begin{array}{c} 0.295^{***} \\ (0.010) \\ 0.546^{***} \\ (0.014) \end{array}$ | $\begin{array}{c} 0.120^{***} \\ (0.009) \\ 0.057^{***} \\ (0.006) \end{array}$ | $\begin{array}{c} 0.193^{***} \\ (0.006) \\ 0.166^{***} \\ (0.006) \end{array}$ | $\begin{array}{c} -0.195^{***} \\ (0.015) \\ -0.137^{***} \\ (0.012) \end{array}$ | $\begin{array}{c} -0.198^{***} \\ (0.013) \\ -0.095^{***} \\ (0.012) \end{array}$ |
| | Diff <i>p</i> -value | [0.000] | [0.000] | [0.003] | [0.002] | [0.000] |
| Capex | CF^{Below} CF^{Above} | $\begin{array}{c} 0.525^{***} \\ (0.014) \\ 0.390^{***} \\ (0.012) \end{array}$ | $\begin{array}{c} 0.030^{***} \\ (0.006) \\ 0.123^{***} \\ (0.009) \end{array}$ | 0.126*** (0.006) 0.209*** (0.006) | $\begin{array}{c} -0.217^{***} \\ (0.014) \\ -0.103^{***} \\ (0.013) \end{array}$ | $\begin{array}{c} -0.102^{***} \\ (0.014) \\ -0.175^{***} \\ (0.011) \end{array}$ |
| | Diff <i>p</i> -value | [0.000] | [0.000] | [0.000] | [0.000] | [0.000] |
| ividends | CF^{Below} CF^{Above} | $\begin{array}{c} 0.487^{***} \\ (0.012) \\ 0.434^{***} \\ (0.013) \end{array}$ | $\begin{array}{c} 0.115^{***} \\ (0.008) \\ 0.063^{***} \\ (0.007) \end{array}$ | $\begin{array}{c} 0.072^{***} \\ (0.005) \\ 0.226^{***} \\ (0.007) \end{array}$ | $\begin{array}{c} -0.193^{***} \\ (0.013) \\ -0.153^{***} \\ (0.013) \end{array}$ | $\begin{array}{c} -0.133^{***} \\ (0.013) \\ -0.123^{***} \\ (0.012) \end{array}$ |
| Di | Diff <i>p</i> -value | [0.003] | [0.000] | [0.000] | [0.033] | [0.602] |
| Debt | CF^{Below} CF^{Above} | $\begin{array}{c} 0.528^{***} \\ (0.012) \\ 0.301^{***} \\ (0.014) \end{array}$ | $\begin{array}{c} 0.066^{***} \\ (0.006) \\ 0.115^{***} \\ (0.009) \end{array}$ | $\begin{array}{c} 0.173^{***} \\ (0.006) \\ 0.182^{***} \\ (0.006) \end{array}$ | $\begin{array}{c} -0.121^{***} \\ (0.010) \\ -0.223^{***} \\ (0.017) \end{array}$ | $\begin{array}{c} -0.112^{***} \\ (0.010) \\ -0.179^{***} \\ (0.015) \end{array}$ |
| | Diff <i>p</i> -value | [0.000] | [0.000] | [0.282] | [0.000] | [0.000] |
| Equity | CF^{Below} CF^{Above} | $\begin{array}{c} 0.425^{***} \\ (0.014) \\ 0.468^{***} \\ (0.012) \end{array}$ | $\begin{array}{c} 0.105^{***} \\ (0.008) \\ 0.060^{***} \\ (0.007) \end{array}$ | $\begin{array}{c} 0.128^{***} \\ (0.005) \\ 0.215^{***} \\ (0.007) \end{array}$ | -0.206^{***} (0.015) -0.136^{***} (0.011) | $\begin{array}{c} -0.136^{***} \\ (0.014) \\ -0.121^{***} \\ (0.011) \end{array}$ |
| | Diff <i>p</i> -value | [0.015] | [0.000] | [0.000] | [0.000] | [0.370] |
| CF | CF ^{Below} CF ^{Above} | $\begin{array}{c} 0.417^{***} \\ (0.013) \\ 0.484^{***} \\ (0.012) \end{array}$ | 0.093*** (0.008) 0.075*** (0.007) | $\begin{array}{c} 0.113^{***} \\ (0.005) \\ 0.199^{***} \\ (0.006) \end{array}$ | -0.204*** (0.015) -0.131*** (0.012) | -0.173*** (0.013) -0.111*** (0.012) |
| | Diff <i>p</i> -value | [0.000] | [0.094] | [0.000] | [0.000] | [0.001] |

| Panel A | A: The | e effects | of | devi | iating | from | the | target | ; |
|---------|--------|-----------|----|------|--------|------|-----|--------|---|
| | | | | | | | | | |

| Panel E | B: Non-line | earities in the o | cash flow sensi | tivities (CF>0 |) | |
|---------------------|--------------------------|----------------------------------------------------------------------------|----------------------------------------------------------------------------|---------------------------------------------------------------------------------|-----------------------------------------------------------------------------------|--------------------------------------------------------------------------------|
| | | $\Delta Cash$ | Capex | Div | ΔD | ΔE |
| Models | Variables | (1) | (2) | (3) | (4) | (5) |
| ithout istraints | CF CF^2 | $\begin{array}{c} 0.511^{***} \\ (0.037) \\ -0.043 \\ (0.035) \end{array}$ | $\begin{array}{c} 0.115^{***} \\ (0.020) \\ -0.028 \\ (0.019) \end{array}$ | $\begin{array}{c} 0.012 \\ (0.018) \\ 0.237^{***} \\ (0.017) \end{array}$ | $\begin{array}{c} -0.100^{***} \\ (0.039) \\ 0.059^{*} \\ (0.036) \end{array}$ | $\begin{array}{c} 0.028 \\ (0.046) \\ -0.185^{***} \\ (0.043) \end{array}$ |
| Cor | $\frac{\mathrm{N}}{R^2}$ | $2,830 \\ 0.196$ | $2,830 \\ 0.175$ | 2,830 0.382 | $2,830 \\ 0.059$ | 2,830 0.066 |
| With astraints | CF CF^2 | $\begin{array}{c} 0.541^{***} \\ (0.037) \\ 0.031 \\ (0.034) \end{array}$ | $\begin{array}{c} 0.117^{***} \\ (0.020) \\ -0.022 \\ (0.019) \end{array}$ | $\begin{array}{c} 0.048^{***} \\ (0.017) \\ 0.325^{***} \\ (0.016) \end{array}$ | $\begin{array}{c} -0.168^{***} \\ (0.036) \\ -0.107^{***} \\ (0.034) \end{array}$ | $\begin{array}{c} -0.125^{***}\\ (0.034)\\ -0.558^{***}\\ (0.032) \end{array}$ |
| Coi | $rac{N}{R^2}$ | $2,830 \\ 0.190$ | $2,830 \\ 0.175$ | $2,830 \\ 0.355$ | $2,830 \\ 0.026$ | $2,830 \\ 0.048$ |

Table 6 The effects of deviating from the target and non-linearities in cash flow sensitivities

Table 7 Alternative estimations of cash flow sensitivities

The table presents the estimation results of a system of equations depicted by Equation (1) relating the uses-of-funds to cash flow and firm characteristics. $\sum \text{Uses}_i = \Delta \text{Cash} + \text{Capex} + \text{Div} + \Delta \text{D} + \Delta \text{E}$. τ is an index of measurement quality of Tobin's q [$0 \ge \tau \le 1$], with zero indicating a poor proxy and one a very good proxy. LR is the Anderson canonical correlations Likelihood ratio test. m2 is a test of second-order autocorrelation in the errors. J is the Hansen test of overidentifying restrictions. All models include control variables (but not reported). The sample consists of listed non-utility and non-financial firms in selected African countries drawn from *Datastream* over the period 2000–2015. All variables used are defined in Appendix A and are winsorised at the lower and upper one percentiles. ***, **, * indicate significance at the one, five, and ten percent levels, respectively.

| | | $\Delta Cash$ | Capex | Div | $\Delta \mathrm{D}$ | ΔE | $\sum \mathrm{Uses}_i$ |
|----------|---------------------------------------------------------------|----------------------------------------------------------------------------------|----------------------------------------------------------------------------------|---------------------------------------------------------------------------------|----------------------------------------------------------------------------------|----------------------------------------------------------------------------------|------------------------|
| | Variables | (1) | (2) | (3) | (4) | (5) | (6) |
| POLS | $\begin{array}{c} \mathrm{CF}_{ijt} \\ q_{ijt-1} \end{array}$ | $\begin{array}{c} 0.284^{***} \\ (0.025) \\ -0.008^{*} \\ (0.005) \end{array}$ | $\begin{array}{c} 0.145^{***} \\ (0.013) \\ 0.003 \\ (0.002) \end{array}$ | $\begin{array}{c} 0.321^{***} \\ (0.033) \\ 0.016^{***} \\ (0.006) \end{array}$ | $\begin{array}{c} -0.085^{***}\\ (0.021)\\ 0.009^{**}\\ (0.004) \end{array}$ | $\begin{array}{c} -0.277^{***} \\ (0.034) \\ -0.011^{*} \\ (0.006) \end{array}$ | 1.100 |
| | $rac{N}{R^2}$ | $5,940 \\ 0.126$ | $5,940 \\ 0.257$ | $5,940 \\ 0.417$ | $5,940 \\ 0.056$ | $5,940 \\ 0.121$ | |
| FE | CF_{ijt} q_{ijt-1} | $\begin{array}{c} 0.415^{***} \\ (0.022) \\ -0.002 \\ (0.003) \end{array}$ | $\begin{array}{c} 0.086^{***} \\ (0.013) \\ 0.004 \\ (0.003) \end{array}$ | $\begin{array}{c} 0.192^{***} \\ (0.023) \\ 0.011^{*} \\ (0.006) \end{array}$ | $\begin{array}{c} -0.120^{***} \\ (0.019) \\ 0.009^{*} \\ (0.005) \end{array}$ | $\begin{array}{c} -0.119^{***} \\ (0.029) \\ 0.000 \\ (0.002) \end{array}$ | 0.923 |
| | $rac{N}{R^2}$ | $5,940 \\ 0.184$ | $5,940 \\ 0.096$ | $5,940 \\ 0.255$ | $5,940 \\ 0.064$ | $5,940 \\ 0.053$ | |
| GMM3 | CF_{ijt} q_{ijt-1} | $\begin{array}{c} -1.871 \\ (13.404) \\ 1.120 \\ (6.584) \end{array}$ | 0.224*** (0.033) -0.063*** (0.008) | $\begin{array}{c} 0.126^{***} \\ (0.022) \\ 0.043^{***} \\ (0.002) \end{array}$ | $\begin{array}{c} -0.804^{***} \\ (0.222) \\ 0.345^{***} \\ (0.090) \end{array}$ | $\begin{array}{c} -0.792 \\ (0.621) \\ 0.330 \\ (0.301) \end{array}$ | 0.070 |
| <u> </u> | $rac{N}{	au}$ | $5,940 \\ 0.272$ | $5,940 \\ 0.324$ | $5,940 \\ 0.802$ | $5,940 \\ 0.279$ | $5,940 \\ 0.284$ | |
| 3MM4 | CF_{ijt} q_{ijt-1} | $\begin{array}{c} 0.847^{***} \\ (0.079) \\ -0.214^{***} \\ (0.026) \end{array}$ | $\begin{array}{c} 0.237^{***} \\ (0.033) \\ -0.069^{***} \\ (0.004) \end{array}$ | $\begin{array}{c} 0.128^{***} \\ (0.021) \\ 0.043^{***} \\ (0.001) \end{array}$ | $\begin{array}{c} -0.664^{***} \\ (0.106) \\ 0.277^{***} \\ (0.028) \end{array}$ | $\begin{array}{c} -0.051 \\ (0.079) \\ -0.033 \\ (0.037) \end{array}$ | 1.922 |
| 0. | $rac{N}{	au}$ | $5,940 \\ 0.284$ | $5,940 \\ 0.319$ | $5,940 \\ 0.810$ | $5,940 \\ 0.281$ | $5,940 \\ 0.182$ | |
| 3MM5 | $\begin{array}{c} \mathrm{CF}_{ijt} \\ q_{ijt-1} \end{array}$ | $\begin{array}{c} 0.676^{***} \\ (0.045) \\ -0.130^{***} \\ (0.009) \end{array}$ | $\begin{array}{c} 0.181^{***} \\ (0.023) \\ -0.042^{***} \\ (0.002) \end{array}$ | $\begin{array}{c} 0.125^{***} \\ (0.022) \\ 0.044^{***} \\ (0.001) \end{array}$ | $\begin{array}{c} -0.485^{***} \\ (0.072) \\ 0.189^{***} \\ (0.015) \end{array}$ | $\begin{array}{c} -0.207^{***} \\ (0.035) \\ 0.043^{***} \\ (0.004) \end{array}$ | 1.662 |
| 0 | $\begin{array}{c} \mathrm{N} \\ 	au \end{array}$ | $5,940 \\ 0.290$ | $5,940 \\ 0.349$ | $5,940 \\ 0.793$ | $5,940 \\ 0.284$ | $5,940 \\ 0.346$ | |

| | | $\Delta Cash$ | Capex | Div | ΔD | ΔE | $\sum \text{Uses}_i$ |
|-------|------------------------------------------------------------|----------------------------------------------------------------------------------|---------------------------------------------------------------------------------|---------------------------------------------------------------------------------|----------------------------------------------------------------------------------|--------------------------------------------------------------------------------|----------------------|
| | Variables | (1) | (2) | (3) | (4) | (5) | (6) |
| -2SLS | $\begin{array}{c} {\rm CF}_{ijt} \\ q_{ijt-1} \end{array}$ | $\begin{array}{c} 0.530^{***} \\ (0.032) \\ -0.032^{***} \\ (0.009) \end{array}$ | $\begin{array}{c} 0.065^{***} \\ (0.017) \\ 0.017^{***} \\ (0.004) \end{array}$ | $\begin{array}{c} 0.151^{***} \\ (0.022) \\ 0.025^{***} \\ (0.004) \end{array}$ | $\begin{array}{c} -0.170^{***} \\ (0.025) \\ 0.032^{***} \\ (0.008) \end{array}$ | $\begin{array}{c} -0.076^{*} \\ (0.040) \\ -0.010 \\ (0.010) \end{array}$ | 0.986 |
| VI | N LR <i>p</i> -value J <i>p</i> -value | 3,783 [0.000] [0.671] | 3,783 [0.000] [0.732] | 3,783 [0.000] [0.001] | 3,783 [0.000] [0.556] | 3,783 [0.000] [0.845] | |
| -GMM | CF_{ijt} q_{ijt-1} | 0.498*** (0.029) -0.020*** (0.007) | $\begin{array}{c} 0.087^{***} \\ (0.029) \\ 0.010 \\ (0.011) \end{array}$ | $\begin{array}{c} 0.169^{***} \\ (0.023) \\ 0.014^{***} \\ (0.005) \end{array}$ | $\begin{array}{c} -0.146^{***} \\ (0.036) \\ 0.022 \\ (0.014) \end{array}$ | $\begin{array}{c} -0.140^{***} \\ (0.045) \\ 0.014 \\ (0.012) \end{array}$ | 1.034 |
| IV | N LR <i>p</i> -value J <i>p</i> -value | $\begin{array}{c} 4,023 \\ [0.000] \\ [0.286] \end{array}$ | $\begin{array}{c} 4,023 \\ [0.000] \\ [0.444] \end{array}$ | $\begin{array}{c} 4,023 \\ [0.000] \\ [0.152] \end{array}$ | $\begin{array}{c} 4,023 \\ [0.000] \\ [0.692] \end{array}$ | $\begin{array}{c} 4,023 \\ [0.000] \\ [0.116] \end{array}$ | |
| 3MM | CF_{ijt} q_{ijt-1} | $\begin{array}{c} 0.510^{***} \\ (0.027) \\ 0.004^{**} \\ (0.002) \end{array}$ | $\begin{array}{c} 0.051^{***} \\ (0.012) \\ 0.000 \\ (0.001) \end{array}$ | $\begin{array}{c} 0.132^{***} \\ (0.019) \\ 0.002 \\ (0.001) \end{array}$ | $\begin{array}{c} -0.201^{***} \\ (0.023) \\ 0.003 \\ (0.002) \end{array}$ | $\begin{array}{c} -0.110^{***} \\ (0.042) \\ 0.008^{*} \\ (0.005) \end{array}$ | 1.003 |
| Ď | N m2 p-value J p-value | 5,301 [0.430] [0.029] | 5,301 [0.553] [0.270] | 5,301 [0.681] [0.943] | 5,301 [0.396] [0.569] | 5,301 [0.016] [0.275] | |
| BMM | CF_{ijt} q_{ijt-1} | 0.388*** (0.025) -0.001 (0.003) | $\begin{array}{c} 0.098^{***} \\ (0.010) \\ 0.002 \\ (0.001) \end{array}$ | $\begin{array}{c} 0.206^{***} \\ (0.053) \\ 0.008 \\ (0.009) \end{array}$ | $\begin{array}{c} -0.102^{***} \\ (0.026) \\ 0.008^{*} \\ (0.004) \end{array}$ | $\begin{array}{c} -0.138^{***} \\ (0.047) \\ 0.000 \\ (0.005) \end{array}$ | 0.922 |
| SC | N m2 p-value J p-value | $5,940 \\ 0.307 \\ [0.187]$ | $5,940 \\ 0.768 \\ [0.815]$ | $5,940 \\ 0.900 \\ [0.295]$ | $5,940 \\ 0.648 \\ [0.186]$ | $5,940 \\ 0.797 \\ [0.853]$ | |
| VAR | CF_{ijt} q_{ijt-1} | $\begin{array}{c} 0.369^{***} \\ (0.040) \\ -0.001 \\ (0.005) \end{array}$ | $\begin{array}{c} 0.075^{***} \\ (0.027) \\ -0.002 \\ (0.003) \end{array}$ | $\begin{array}{c} 0.166^{***} \\ (0.028) \\ 0.002 \\ (0.002) \end{array}$ | $\begin{array}{c} -0.238^{***} \\ (0.043) \\ -0.000 \\ (0.005) \end{array}$ | $\begin{array}{c} -0.078 \\ (0.069) \\ -0.007 \\ (0.010) \end{array}$ | 0.910 |
| -P | N J J <i>p-value</i> | 4,662 87.120 [0.000] | 4,662 87.120 [0.000] | 4,662 87.120 [0.000] | 4,662 87.120 [0.000] | 4,662 87.120 [0.000] | |

Table 7 Alternative estimations of cash flow sensitivities (continued)

| sensitivities |
|-----------------------|
| flow |
| cash |
| in |
| variations |
| Time |
| Ø |
| Table |

The table presents the estimation results of a system of equations depicted by Equation (1) relating the uses-of-funds to cash flow and firm characteristics. The correlation of residuals across the system of equations is not reported for brevity. All models include control variables (but not reported). The sample consists of listed non-utility and non-financial firms in selected African countries drawn from *Datastream* over the period 2000–2015. All variables used are defined in Appendix A and are winsorised at the lower and upper one percentiles. ***, **, * indicate significance at the one, five, and ten percent levels, respectively.

| | | | | SUREG | | | | | FM | | |
|------|-----|--------------------------|--------------------------|--------------------------|--------------------------|----------------------------|--------------------------|--------------------------|--------------------------|---------------------------|---------------------------|
| | z | $\Delta \mathrm{Cash}$ | Capex | Div | ΔD | ΔE | $\Delta Cash$ | Capex | Div | ΔD | ΔE |
| Year | | (1) | (2) | (3) | (4) | (5) | (9) | (2) | (8) | (6) | (10) |
| 2000 | 171 | 0.406^{***} | 0.068 | 0.110^{***} | 0.045 | -0.461^{***} | 0.502^{***} | 0.081^{**} | 0.184^{***} | -0.092* | -0.236^{***} |
| 2001 | 231 | (0.067) 0.250^{***} | $(0.049) \\ 0.025$ | $(0.033) \\ 0.417^{***}$ | $(0.098) \\ -0.106^{**}$ | (0.108) - 0.202^{***} | $(0.064) \\ 0.139^{**}$ | $(0.040) \\ 0.153^{***}$ | $(0.022) \\ 0.489^{***}$ | $(0.049) \\ -0.097^{***}$ | $(0.071) - 0.336^{***}$ |
| | | (0.055) | (0.035) | (0.026) | (0.052) | (0.048) | (0.059) | (0.034) | (0.059) | (0.030) | (0.068) |
| 2002 | 278 | 0.533^{***} | -0.020 | 0.052^{***} | -0.355*** | -0.081^{*} | 0.582^{***} | -0.129^{***} | 0.083^{***} | -0.066 | 0.156^{**} |
| 2003 | 310 | (0.591^{***}) | (0.031)-0.125*** | $(0.011) -0.034^{*}$ | $(0.001) -0.467^{***}$ | $(0.040) - 0.102^{**}$ | (0.000) 0.287^{***} | (0.041) | (0.032) 0.078** | (0.030) -0.103 | (0.000) - 0.302^{***} |
| | 0 | (0.064) | (0.040) | (0.019) | (0.072) | (0.052) | (0.090) | (0.045) | (0.034) | (0.091) | (0.087) |
| 2004 | 331 | 0.387^{***} | 0.119*** (0.094) | 0.042 | -0.221*** | -0.231*** (0.037) | 0.335^{***} | 0.145*** (0.037) | 0.062 | -0.160*** (0.049) | -0.326*** (0.061) |
| 2005 | 340 | 0.132^{***} | 0.031 | 0.234^{***} | -0.350^{***} | -0.253^{***} | 0.084^{**} | 0.196^{***} | 0.405^{**} | -0.157^{**} | -0.337^{*} |
| 0000 | 700 | (0.051) | (0.032) | (0.037) | (0.059) | (0.056) | (0.042) | (0.035) | (0.170) | (0.075) | (0.202) |
| 2006 | 365 | 0.560*** (0.050) | 0.042 (0.020) | (0.121^{***}) | -0.216*** | -0.061 | 0.364^{***} | (0.205^{***}) | (0.162^{***}) | -0.230*** | -0.203*** |
| 2007 | 392 | 0.449^{***} | 0.158^{***} | 0.186^{***} | (0.032) | -0.239^{***} | 0.252^{***} | 0.180^{***} | 0.394^{***} | 0.000 | -0.443^{***} |
| | | (0.065) | (0.030) | (0.028) | (0.062) | (0.056) | (0.042) | (0.039) | (0.038) | (0.036) | (0.077) |
| 2008 | 401 | 0.474^{***} | 0.069^{**} | 0.259^{***} | -0.039 | -0.160^{***} | 0.429^{***} | 0.058^{**} | 0.444^{***} | -0.142^{***} | -0.326^{***} |
| 2009 | 444 | (0.049) 0.453^{***} | $(0.02l) \\ 0.142^{***}$ | (0.022) 0.204^{***} | $(0.041) -0.084^{*}$ | (0.049) - 0.117^{**} | (0.053^{***}) | (0.029) 0.184^{***} | (0.083) 0.282^{***} | (ccn.n) -0.009 | $(0.100) - 0.219^{***}$ |
| | | (0.042) | (0.035) | (0.022) | (0.050) | (0.047) | (0.050) | (0.048) | (0.047) | (0.053) | (0.074) |
| 2010 | 487 | 0.527^{***} | 0.105^{***} | 0.144^{***} | -0.130^{**} | -0.095^{*} | 0.386^{***} | 0.058^{*} | 0.296^{***} | -0.070^{*} | -0.308^{***} |
| 2011 | 508 | (0.041) 0.309^{***} | $(0.036) \\ 0.114^{***}$ | $(0.021) \\ 0.126^{***}$ | $(0.056) -0.242^{***}$ | (0.050) - 0.209^{***} | $(0.038) \\ 0.224^{***}$ | $(0.031) \\ 0.117^{***}$ | $(0.033) \\ 0.267^{***}$ | $(0.038) -0.179^{***}$ | $(0.058) \\ -0.133^{**}$ |
| | | (0.037) | (0.026) | (0.021) | (0.040) | (0.034) | (0.039) | (0.035) | (0.040) | (0.031) | (0.067) |
| 2012 | 477 | 0.477^{***} | 0.146^{***} | 0.131^{***} | -0.265^{***} | 0.019 | 0.240^{***} | 0.266^{***} | 0.289^{***} | -0.184^{***} | -0.147^{**} |
| 2013 | 447 | (0.045) 0.401^{***} | (0.026) 0.133^{***} | $(0.020) \\ 0.121^{***}$ | $(0.046) -0.220^{***}$ | (0.045) - 0.125^{***} | $(0.049) \\ 0.307^{***}$ | (0.042) 0.175^{***} | $(0.043) \\ 0.265^{***}$ | $(0.038) -0.145^{***}$ | (0.070) -0.111 |
| | | (0.042) | (0.030) | (0.021) | (0.047) | (0.037) | (0.052) | (0.042) | (0.034) | (0.041) | (0.073) |
| 2014 | 412 | 0.484^{***} | 0.110^{***} | 0.217^{***} | -0.144^{**} | -0.045 | 0.367^{***} | 0.183^{***} | 0.227^{***} | -0.088 | -0.035 |
| 2015 | 346 | (0.063) 0.389 $***$ | (0.037) | $(0.021) \\ 0.202^{***}$ | (0.062) -0.171*** | (0.051) -0.186*** | $(0.061) \\ 0.378^{***}$ | (0.035)0.279 $***$ | $(0.032) \\ 0.161^{***}$ | (0.060) | (0.097)-0.217*** |
| 0107 | 010 | (0.055) | (0.039) | (0.030) | (0.054) | (0.053) | (0.083) | (0.054) | (0.052) | (0.052) | (0.064) |

| (continued | sensitivities | flow | cash | in | variations | Time | × | ble |
|------------|---------------|------|------|----|------------|-----------------------|---|-----|

Table 9 Cash flow sensitivities across countries and industries

The table presents the estimation results of a system of equations depicted by Equation (1) relating the uses-of-funds to cash flow and firm characteristics. Panel A presents the results for the sub-samples (FIC) of other countries (Others – Egypt, Ivory Coast, Kenya, Morocco, Nigeria and Tunisia) and South Africa (S. Africa). Panel B presents estimation results across the industries (SIC); namely, Industrials (IND), Health Care (HC), Consumer Goods and Services (CG&S), Technology and Telecommunications (T&T) and Others (Basic Materials and Oil & Gas). $\sum Uses_i = \Delta Cash + Capex + Div + \Delta D + \Delta E$. All models include control variables and the lagged use-of-funds (but not reported). The correlation of residuals across the system of equations is not reported for brevity. The sample consists of listed non-utility and non-financial firms in selected African countries drawn from *Datastream* over the period 2000–2015. All variables used are defined in Appendix A and are winsorised at the lower and upper one percentiles. ***, **, * indicate significance at the one, five, and ten percent levels, respectively.

| | | $\Delta Cash$ | Capex | Div | ΔD | ΔE | $\sum \text{Uses}_i$ |
|----------------|--------------------------|-------------------------------------------------------|-------------------------------------------------------|-------------------------------------------------------|---------------------------|---------------------------|----------------------|
| FIC | Variables | (1) | (2) | (3) | (4) | (5) | (6) |
| hers | CF_{ijt} | $\begin{array}{c} 0.467^{***} \\ (0.011) \end{array}$ | $\begin{array}{c} 0.128^{***} \\ (0.009) \end{array}$ | $\begin{array}{c} 0.131^{***} \\ (0.005) \end{array}$ | -0.223^{***} (0.011) | -0.051^{***} (0.010) | 0.996 |
| Ot | $\frac{\mathrm{N}}{R^2}$ | $1,142 \\ 0.30$ | $1,142 \\ 0.21$ | $1,142 \\ 0.31$ | $1,142 \\ 0.11$ | $1,142 \\ 0.04$ | |
| frica | CF_{ijt} | $\begin{array}{c} 0.460^{***} \\ (0.013) \end{array}$ | 0.057^{***} (0.007) | $\begin{array}{c} 0.160^{***} \\ (0.007) \end{array}$ | -0.158^{***} (0.014) | -0.165^{***} (0.013) | 0.987 |
| \mathbf{S} . | $\frac{\mathrm{N}}{R^2}$ | $4,798 \\ 0.23$ | $4,798 \\ 0.19$ | $4,798 \\ 0.33$ | $4,798 \\ 0.06$ | $4,798 \\ 0.03$ | |
| Diff (| CF p-value | [0.660] | [0.000] | [0.001] | [0.000] | [0.000] | |

| Panel A: Cash flow sensitivities across count | ries |
|-----------------------------------------------|------|
|-----------------------------------------------|------|

| Panel B: | Cash flow | sensitivities | across | industries |
|----------|-----------|----------------|---------|------------|
| I and D. | Cash now | 30113101 10103 | aci 055 | mausure |

| SIC | Variables | (1) | (2) | (3) | (4) | (5) | (6) |
|---------|-------------------------------------------------|-------------------------------------------------------|-------------------------------------------------------|-------------------------------------------------------|---------------------------|---------------------------|-------|
| t&S | CF_{ijt} | $\begin{array}{c} 0.505^{***} \\ (0.020) \end{array}$ | 0.056^{***} (0.011) | $\begin{array}{c} 0.131^{***} \\ (0.009) \end{array}$ | -0.238^{***} (0.022) | -0.070^{***} (0.021) | 0.992 |
| GG | $egin{array}{c} N \ R^2 \end{array}$ | 2,201 0.269 | 2,201 0.183 | 2,201 0.406 | 2,201 0.042 | 2,201 0.018 | |
| C | CF_{ijt} | 0.260^{**} (0.131) | 0.060^{*} (0.034) | -0.002 (0.048) | -0.393^{***} (0.111) | -0.289^{***} (0.074) | 0.990 |
| Щ | $rac{\mathrm{N}}{R^2}$ | $212 \\ 0.269$ | $212 \\ 0.195$ | $212 \\ 0.037$ | $212 \\ 0.407$ | $212 \\ 0.103$ | |
| IND | CF_{ijt} | $\begin{array}{c} 0.582^{***} \\ (0.026) \end{array}$ | 0.093^{***} (0.019) | $\begin{array}{c} 0.033^{***} \\ (0.011) \end{array}$ | -0.230^{***} (0.027) | -0.063^{**} (0.025) | 0.998 |
| | $egin{array}{c} { m N} \\ { m R}^2 \end{array}$ | $1,684 \\ 0.252$ | $1,684 \\ 0.200$ | $1,684 \\ 0.297$ | $1,684 \\ 0.107$ | $1,684 \\ 0.050$ | |
| Others | CF_{ijt} | $\begin{array}{c} 0.280^{***} \\ (0.020) \end{array}$ | $\begin{array}{c} 0.121^{***} \\ (0.013) \end{array}$ | $\begin{array}{c} 0.288^{***} \\ (0.013) \end{array}$ | -0.064^{***} (0.024) | -0.248^{***} (0.022) | 0.989 |
| | $rac{N}{R^2}$ | $1,371 \\ 0.292$ | $1,371 \\ 0.305$ | $1,371 \\ 0.457$ | $1,371 \\ 0.099$ | $1,371 \\ 0.139$ | |
| ζŢ | $\overline{\mathrm{CF}_{ijt}}$ | $\begin{array}{c} 0.511^{***} \\ (0.045) \end{array}$ | $\begin{array}{c} 0.088^{***} \\ (0.022) \end{array}$ | $\begin{array}{c} 0.073^{***} \\ (0.020) \end{array}$ | -0.024 (0.037) | -0.304^{***} (0.039) | 0.992 |
| T_{2} | $rac{N}{R^2}$ | 472 0.236 | 472 0.224 | 472 0.400 | 472 0.066 | 472 0.006 | |

Appendix A Variable definitions

The table lists the definitions of all variables used. All firm-level data is from $Thomson \ DataStream$, and macroeconomic variables are from The World Bank.

| Variable | Definition |
|---------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| $\Delta Cash$ | Changes in cash and equivalent (WC02001)-to-total assets (WC02999). |
| Capex | Physical capital investments (WC04601)-to-total assets. |
| Div | Dividends (WC05376)-to-total assets. |
| ΔD | Changes in total debt (WC03251+WC03051)-to-total assets. |
| ΔE | Changes in equity (WC03480)-to-total assets. |
| \mathbf{CF} | Net cash flow from operating activities (NOCF)-to-total assets. |
| | NOCF=Net income (WC04001)+Depreciation, depletion & amortization (WC04051) |
| | +Deferred income taxes & investments (WC04101)+Total other cash flow (WC04151) |
| | +Extraordinary items (WC04225)+Funds from other operating activities (WC04831). |
| $\rm CF1$ | Operating income (WC01250) plus depreciation (WC04051)-to-total assets. |
| $\rm CF2$ | Net income before extraordinary Items (WC01551) $plus$ depreciation (WC04051)-to- |
| | total assets. |
| Cash | Cash and equivalent (WC02001)-to-total assets. |
| Debt | Total debt ($WC03251+WC03051$)-to-total assets. |
| q | Market value of equity (MV) plus total debt (WC03251+WC03051)-to-total assets. |
| SG | Sales growth (WC01001) |
| Size | Log of total assets (WC02999). |
| PPE | Property, plant and equipment (WC02501)-to-total assets. |
| LogAge | The current year less the first year that the firm appears in the database. |
| WW Index | $-0.091 	imes rac{Cash\ Flow}{Total\ Assets} - 0.062 	imes DivDummy + 0.021 	imes rac{Total\ debt}{Total\ Assets}$ |
| | $-0.044 \times Size + 0.102 \times IndustrySalesGrowth - 0.035 \times SG$ |
| | The WW Index is based on Whited and Wu (2006). |
| HP Index | $-0.737 \times Size + 0.043 \times Size^2 - 0.040 * Age.$ |
| | The HP Index is based on Hadlock and Pierce (2010). |
| KZ Index | $-1.002 \times \frac{Cash}{Total} \frac{Flow}{Assets} + 0.283 \times \frac{Total}{Total} \frac{debt}{Assets} - 39.368 \times \frac{Dividends}{Total} - 1.315 \times \frac{Cash}{Total} \frac{Cash}{Assets} + 0.283 \times \frac{Total}{Total} \frac{debt}{Assets} - 39.368 \times \frac{Dividends}{Total} - 1.315 \times \frac{Cash}{Total} \frac{Cash}{Assets} + 0.283 \times \frac{Total}{Total} \frac{debt}{Assets} - 39.368 \times \frac{Dividends}{Total} - 1.315 \times \frac{Cash}{Total} \frac{Cash}{Assets} + 0.283 \times \frac{Total}{Total} \frac{debt}{Assets} - 39.368 \times \frac{Dividends}{Total} - 1.315 \times \frac{Cash}{Total} \frac{Cash}{Assets} + 0.283 \times \frac{Dividends}{Total} \frac{Cash}{Assets} - 1.315 \times \frac{Cash}{Total} \frac{Cash}{Assets} + 0.283 \times \frac{Dividends}{Total} \frac{Cash}{Assets} - 1.315 \times \frac{Cash}{Total} \frac{Cash}{Assets} + 0.283 \times \frac{Dividends}{Total} \frac{Cash}{Assets} - 1.315 \times \frac{Cash}{Total} \frac{Cash}{Assets} + 0.283 \times \frac{Dividends}{Total} \frac{Cash}{Assets} - 1.315 \times \frac{Cash}{Total} \frac{Cash}{Assets} + 0.283 \times \frac{Dividends}{Total} \frac{Cash}{Assets} - 1.315 \times \frac{Cash}{Total} \frac{Cash}{Assets} + 0.283 \times \frac{Dividends}{Total} \frac{Cash}{Assets} - 1.315 \times \frac{Cash}{Total} \frac{Cash}{Assets} + 0.283 \times \frac{Dividends}{Total} \frac{Cash}{Assets} - 1.315 \times \frac{Cash}{Total} \frac{Cash}{Assets} + 0.283 \times \frac{Dividends}{Total} \frac{Cash}{Assets} - 1.315 \times \frac{Cash}{Total} \frac{Cash}{Assets} + 0.283 \times \frac{Dividends}{Total} \frac{Cash}{Assets} - 1.315 \times \frac{Cash}{Total} \frac{Cash}{Assets} + 0.283 \times \frac{Dividends}{Total} \frac{Cash}{Assets} - 1.315 \times \frac{Cash}{Total} \frac{Cash}{Assets} + 0.283 \times \frac{Dividends}{Total} \frac{Cash}{Assets} - 1.315 \times \frac{Cash}{Total} \frac{Cash}{Assets} - 1.315 \times \frac{Cash}{Total} \frac{Cash}{Assets} + 0.283 \times \frac{Dividends}{Total} \frac{Cash}{Assets} - 1.315 \times \frac{Cash}{Total} \frac{Cash}{Assets} - 1.315 \times $ |
| | The KZ Index is based on Kaplan and Zingales (1997). |
| GDPG | GDP growth (annual %). |
| IRS | Interest rate spread (lending rate minus deposit rate $\%$). |
| INF | Inflation, consumer prices (annual %). |
| PVTCREDIT | Domestic credit to the private sector by banks ($\%$ of GDP). |
| STMKTCAP | Stock market capitalisation-to-GDP (% of GDP). |

| conditions |
|-----------------|
| macroeconomic |
| and |
| sensitivities |
| flow |
| \mathbf{Cash} |
| р |
| pendix |
| Ap |

The table presents the estimation results of a system of equations depicted by Equation (1) relating the uses-of-funds to cash flow, firm characteristics and macroeconomic factors. The correlation of residuals across the system of equations is not reported for brevity. The sample consists of listed non-utility and non-financial firms in selected African countries drawn from *Datastream* over the period 2000–2015. All variables used are defined in Appendix A and are winsorised at the lower and upper one percentiles. ***, **, * indicate significance at the one, five, and ten percent levels, respectively.

| | | 2 | | | | | | | | |
|-----------------------------------------|------------------------|--------------------------------------------|----------------------|---------------------|---------------------|------------------------|--------------------------------------------|----------------------|----------------------|---------------------|
| | | Models | s without con | straints | | | Mode | ls with const. | raints | |
| | $\Delta \mathrm{Cash}$ | Capex | Div | ΔD | ΔE | $\Delta \mathrm{Cash}$ | Capex | Div | ΔD | $\Delta \mathrm{E}$ |
| Variables | (1) | (2) | (3) | (4) | (5) | (9) | (2) | (8) | (6) | (10) |
| CF_{ijt} | 0.468^{***} | 0.080^{***} | 0.175^{***} | -0.145^{***} | -0.099*** | 0.471^{***} | 0.080^{***} | 0.180^{***} | -0.153^{***} | -0.116^{***} |
| , , | (0.014) | (0.008) | (0.008) | (0.016) | (0.019) | (0.014) | (0.008) | (0.007) | (0.015) | (0.014) |
| q_{iit-1} | 0.000 | 0.002^{***} | 0.008^{***} | 0.005^{***} | 0.003^{**} | -0.000 | 0.002^{***} | 0.007^{***} | 0.005^{***} | 0.004^{***} |
| | (0.001) | (0.001) | (0.001) | (0.001) | (0.002) | (0.001) | (0.001) | (0.001) | (0.001) | (0.001) |
| SG_{ijt-1} | 0.005 | -0.002 | 0.018^{***} | 0.001 | -0.007 | 0.002 | -0.02 | 0.014^{***} | 0.008 | 0.007 |
| | (0.006) | (0.004) | (0.003) | (0.007) | (0.008) | (0.006) | (0.004) | (0.003) | (0.006) | (0.006) |
| $\mathbf{S}\mathbf{IZ}\mathbf{e}_{ijt}$ | -0.016*** | -0.012*** | 0.011*** | -0.040*** | -0.024*** | -0.020^{***} | -0.012*** | 0.005*** | -0.028^{+++} | 0.000 |
| $PPF_{\mathbb{C}}$ | (U.UU4) 0 144*** | (0.002) -0 106*** | (0.002) -0.121*** | (0.004) -0.042** | (0.000) 0 167*** | (0.004) 0 163*** | (0.002) -0 105*** | (0.002) -0.094*** | (0.004) -0.095*** | (0.004) 0.059*** |
| | (0.018) | (0.011) | (0.010) | (0.021) | (0.024) | (0.018) | (0.011) | (0.00) | (0.019) | (0.018) |
| $\Delta \mathrm{Cash}_{ijt-1}$ | -0.255^{***} | 0.004 | 0.032^{***} | 0.021 | -0.067^{***} | -0.239^{***} | 0.005 | 0.055^{***} | -0.023^{*} | -0.156^{***} |
| \$ | (0.013) | (0.007) | (0.007) | (0.014) | (0.017) | (0.013) | (0.007) | (0.006) | (0.013) | (0.012) |
| $\operatorname{Capex}_{ijt-1}$ | -0.124^{***} | 0.375^{***} | 0.035^{**} | 0.268^{***} | -0.087*** | -0.134^{***} | 0.374^{***} | 0.021 | 0.294^{***} | -0.033 |
| 5 | (0.025) | (0.015) | (0.014) | (0.028) | (0.033) | (0.025) | (0.015) | (0.013) | (0.026) | (0.024) |
| ${ m Div}_{ijt-1}$ | -0.144^{***} | 0.021 | 0.178^{***} | 0.222^{***} | -0.154^{***} | -0.143^{***} | 0.021 | 0.180^{***} | 0.218^{***} | -0.160^{***} |
| 5 | (0.026) | (0.015) | (0.014) | (0.029) | (0.034) | (0.026) | (0.015) | (0.013) | (0.027) | (0.025) |
| $\Delta \mathrm{D}_{ijt-1}$ | 0.028^{**} | -0.014^{*} | -0.082*** | -0.098*** | 0.097^{***} | 0.034^{**} | -0.014^{*} | -0.073*** | -0.115^{***} | 0.062^{***} |
| | (0.013) | (0.008) | (0.007) | (0.015) | (0.017) | (0.013) | (0.008) | (0.007) | (0.014) | (0.013) |
| $\Delta \mathrm{E}_{ijt-1}$ | 0.063^{***} | -0.043^{***} | -0.027^{***} | 0.007 | 0.030^{*} | 0.067^{***} | -0.043^{***} | -0.021^{***} | -0.004 | 0.007 |
| | (0.012) | (0.007) | (10.007) | (0.013) | (0.016) | (0.012) | (0.007) | (0.006) | (0.012) | (0.011) |
| unr u_{ijt-1} | -0.110 | (0.073) (0.073) | (0.060) | 0.403 (0.149) | -0.016 (0.164) | -0.117 (0 196) | 0.390 (0.073) | (0.063) | 0.400 (0.131) | -0.009 |
| IRS_{iit-1} | 0.035 | -0.026 | 0.119^{**} | 0.094 | -0.018 | 0.030 | -0.026 | 0.112^{**} | 0.107 | 0.009 |
| - | (0.091) | (0.052) | (0.050) | (0.102) | (0.118) | (0.090) | (0.052) | (0.045) | (0.094) | (0.087) |
| INF_{ijt-1} | 0.451^{*} | -0.408^{***} | -0.386^{***} | 0.323 | 0.397 | 0.551^{**} | -0.405^{***} | -0.245^{*} | 0.053 | -0.153 |
| 5 | (0.272) | (0.157) | (0.148) | (0.303) | (0.351) | (0.269) | (0.157) | (0.135) | (0.280) | (0.260) |
| $^{ m N}_{R^2}$ | $4,957 \\ 0.25$ | $\begin{array}{c} 4.957\\ 0.19\end{array}$ | $4,957 \\ 0.26$ | $4,957 \\ 0.07$ | $4,957 \\ 0.04$ | $4,957 \\ 0.25$ | $\begin{array}{c} 4,957\\ 0.19\end{array}$ | $4,957 \\ 0.25$ | $4,957 \\ 0.07$ | $4,957 \\ 0.02$ |

Appendix C Cash flow and uses-of-funds across industries

The table presents time-series summary statistics and pairwise correlations between the proxies of cash flow. The sample consists of listed non-utility and non-financial firms in selected African countries drawn from *Datastream* over the period 2000–2015. $\sum \text{Uses}_i = \Delta \text{Cash} + \text{Capex} + \text{Div} + \Delta \text{D} + \Delta \text{E}$. All variables used are defined in Appendix A and are winsorised at the lower and upper one percentiles. ***, **, * indicate significance at the one, five, and ten percent levels, respectively.

| | | Variables | \mathbf{CF} | $\Delta Cash$ | Capex | Div | ΔDebt | $\Delta Equity$ | CF vs Uses |
|-----|--------|--------------------------------|-----------------------------------------------------------------|--------------------------------------------------------|--------------------------------------------------------|--------------------------------------------------------|--------------------------------------------------------|----------------------------|------------------|
| # | SIC | Metric | (1) | (2) | (3) | (4) | (5) | (6) | (7) |
| (1) | CG&S | Mean Median Std.Dev N | $\begin{array}{c} 0.163 \\ 0.152 \\ 0.149 \\ 2,201 \end{array}$ | $\begin{array}{c} 0.018 \\ 0.006 \\ 0.037 \end{array}$ | $\begin{array}{c} 0.072 \\ 0.060 \\ 0.056 \end{array}$ | $\begin{array}{c} 0.062 \\ 0.049 \\ 0.048 \end{array}$ | $\begin{array}{c} 0.015 \\ 0.000 \\ 0.064 \end{array}$ | -0.074 -0.073 -0.033 | -0.048 -0.036 |
| (2) | HC | Mean Median Std.Dev N | $\begin{array}{c} 0.149 \\ 0.130 \\ 0.139 \\ 212 \end{array}$ | $\begin{array}{c} 0.037 \\ 0.021 \\ 0.020 \end{array}$ | $0.056 \\ 0.050 \\ 0.079$ | $\begin{array}{c} 0.048 \\ 0.041 \\ 0.043 \end{array}$ | $\begin{array}{c} 0.064 \\ 0.003 \\ 0.024 \end{array}$ | -0.033 -0.040 -0.056 | 0.039 -0.019 |
| (3) | IND | Mean Median Std.Dev N | $\begin{array}{c} 0.139 \\ 0.120 \\ 0.152 \\ 1,684 \end{array}$ | $\begin{array}{c} 0.020 \\ 0.008 \\ 0.011 \end{array}$ | $\begin{array}{c} 0.079 \\ 0.057 \\ 0.096 \end{array}$ | $\begin{array}{c} 0.043 \\ 0.026 \\ 0.053 \end{array}$ | $\begin{array}{c} 0.024 \\ 0.002 \\ 0.013 \end{array}$ | -0.056 -0.054 -0.045 | -0.035 -0.023 |
| (4) | Others | Mean Median Std.Dev N | $\begin{array}{c} 0.152 \\ 0.125 \\ 0.204 \\ 1,371 \end{array}$ | $\begin{array}{c} 0.011 \\ 0.006 \\ 0.020 \end{array}$ | $\begin{array}{c} 0.096 \\ 0.084 \\ 0.076 \end{array}$ | $\begin{array}{c} 0.053 \\ 0.021 \\ 0.069 \end{array}$ | $\begin{array}{c} 0.013 \\ 0.001 \\ 0.016 \end{array}$ | -0.045 -0.035 -0.080 | -0.04 -0.02 |
| (5) | T&T | Mean Median Std.Dev N | $\begin{array}{c} 0.204 \\ 0.200 \\ 0.000 \\ 472 \end{array}$ | $\begin{array}{c} 0.020 \\ 0.015 \\ 0.018 \end{array}$ | $0.076 \\ 0.053 \\ 0.000$ | $\begin{array}{c} 0.069 \\ 0.046 \\ 0.000 \end{array}$ | $0.016 \\ 0.000 \\ 0.406$ | -0.080 -0.070 0.000 | -0.025 0.016 |

Panel A: Basic statistics across industries

| Panel B: | Differences | s across ind | ustries | | | | |
|------------|---------------------------|-------------------------------------------------|-------------------------------------------------|-------------------------------------|--------------------------------------------------------------------|-------------------------------------------------|-------------------------------------|
| | Variables | CF | $\Delta Cash$ | Capex | Div | ΔDebt | $\Delta Equity$ |
| SIC | Metric | (1) | (2) | (3) | (4) | (5) | (6) |
| (1) vs (2) | Mean Median Std.Dev | 0.014** 0.022* 0.010** | -0.019*** -0.015*** 0.017*** | 0.016*** 0.010*** -0.023*** | $\begin{array}{c} 0.014^{***} \\ 0.008^{***} \\ 0.005 \end{array}$ | -0.049*** -0.003 0.040*** | -0.041*** -0.033*** 0.023 |
| (1) vs (3) | Mean Median Std.Dev | 0.024^{***} 0.043^{***} 0.011^{***} | -0.002 0.010 0.007 | -0.007*** 0.015*** -0.024*** | 0.019^{***} 0.036^{***} 0.009^{***} | -0.009*** 0.013*** 0.002*** | -0.018*** -0.020*** -0.029*** |
| (1) vs (4) | Mean Median Std.Dev | 0.011*** 0.038*** -0.041*** | 0.007*** 0.012*** -0.002*** | -0.024*** -0.012*** -0.004*** | 0.009*** 0.041*** -0.007*** | 0.002 0.014 -0.001 | -0.029*** -0.039*** 0.006*** |
| (1) vs (5) | Mean Median Std.Dev | -0.041*** -0.037*** 0.163*** | -0.002 0.003 0.000 | -0.004 0.019 0.072 | -0.007** 0.016** 0.062** | -0.001 0.015 -0.391 | 0.006 -0.004 -0.074 |
| (2) vs (3) | Mean Median Std.Dev | 0.010 0.010 -0.013 | 0.017^{***} 0.013^{***} 0.009^{***} | -0.023*** -0.007*** -0.017*** | $\begin{array}{c} 0.005 \\ 0.015 \\ -0.010 \end{array}$ | 0.040^{***} 0.001^{***} 0.011^{***} | 0.023*** 0.014*** -0.011*** |
| (2) vs (4) | Mean Median Std.Dev | -0.003 0.005 -0.065 | 0.026^{***} 0.015^{***} 0.000^{***} | -0.040*** -0.034*** 0.003*** | -0.005 0.020 -0.026 | 0.051^{***} 0.002^{***} 0.008^{***} | $0.012 \\ -0.005 \\ 0.024$ |
| (2) vs (5) | Mean Median Std.Dev | -0.055*** -0.070*** 0.139*** | 0.017** 0.006** 0.002** | -0.020*** -0.003*** 0.079*** | -0.021*** -0.005*** 0.043*** | 0.048*** 0.003*** -0.382*** | 0.047*** 0.030*** -0.056*** |
| (3) vs (4) | Mean Median Std.Dev | -0.013*** -0.005*** -0.052*** | 0.009*** 0.002*** -0.009*** | -0.017*** -0.027*** 0.020*** | -0.010*** 0.005*** -0.016*** | 0.011*** 0.001*** -0.003*** | -0.011*** -0.019*** 0.035*** |
| (3) vs (5) | Mean Median Std.Dev | -0.065*** -0.080*** 0.120*** | 0.000 -0.007 -0.010 | $0.003 \\ 0.004 \\ 0.057$ | -0.026*** -0.020*** 0.026*** | 0.008* 0.002* -0.404* | 0.024*** 0.016*** -0.054*** |
| (4) vs (5) | Mean Median Std.Dev | -0.052*** -0.075*** 0.204*** | -0.009** -0.009** 0.002** | 0.020*** 0.031*** 0.076*** | -0.016*** -0.025*** 0.069*** | -0.003 0.001 -0.390 | 0.035*** 0.035*** -0.080*** |

Appendix C Cash flow and uses-of-funds across industries (continued)

| flow |
|---------------|
| cash |
| \mathbf{of} |
| proxies |
| ternative |
| Al |
| Ω |
| Appendix |

The table presents the estimation results of a system of equations depicted by Equation (1) relating the uses-of-funds to cash flow and firm characteristics. Panel A presents the estimates based on CF1. Panel B presents the estimation results based on CF2. CF1 is operating income *plus* depreciation-to-total assets. CF2 is net income before extraordinary items *plus* depreciation-to-total assets. All models include control variables and the lagged use-of-funds (but not reported). The correlation of residuals across the system of equations is not reported for brevity. The sample consists of listed non-utility and non-financial firms in selected African countries drawn from *Datastream* over the period 2000–2015. All variables used are defined in Appendix A and are winsorised at the lower and upper one percentiles. ***, **, indicate significance at the one, five, and ten percent levels, respectively.

| Fanel A: | CFI | | | | | | | | | |
|-------------------------------------------------------------------|--------------------------|--------------------------|--------------------------|---------------------------|---------------------------|-------------------------|--------------------------|-----------------------------------------------------------------|----------------------------------------------------------------------|---------------------------|
| | | Model | ls without con | ıstraints | | | Mode | els with const | traints | |
| | $\Delta Cash$ | Capex | Div | ΔD | ΔE | $\Delta Cash$ | Capex | Div | ΔD | ΔE |
| Variables | (1) | (2) | (3) | (4) | (5) | (9) | (2) | (8) | (6) | (10) |
| CF_{ijt} | 0.177*** | 0.067*** | 0.166^{***} | 0.028** | -0.305^{***} | 0.230^{***} | 0.068*** | 0.204^{***} | -0.062^{***} | -0.435^{***} |
| q_{ijt-1} | (0.001) | 0.002^{***} (0.001) | (0.001) | (0.001) | (0.001) | (0.001) | (0.001) | (0.001) | (0.003^{***}) | (0.001) |
| $\begin{array}{c} \text{Controls} \\ \text{N} \\ R^2 \end{array}$ | Yes 5,940 0.118 | Yes 5,940 0.185 | Yes 5,940 0.307 | Yes 5,940 0.050 | Yes 5,940 0.112 | Yes 5,940 0.112 | Yes 5,940 0.185 | $\begin{array}{c} \mathrm{Yes} \\ 5,940 \\ 0.298 \end{array}$ | Yes 5,940 0.035 | Yes 5,940 0.087 |
| Panel B: | CF2 | | | | | | | | | |
| | | Model | ls without con | ıstraints | | | Mode | els with const | traints | |
| | $\Delta \mathrm{Cash}$ | Capex | Div | ΔD | ΔE | $\Delta \mathrm{Cash}$ | Capex | Div | ΔD | ΔE |
| Variables | (1) | (2) | (3) | (4) | (5) | (9) | (7) | (8) | (6) | (10) |
| CF_{ijt} | 0.230^{***} (0.011) | 0.068^{***} (0.006) | 0.204^{***} (0.005) | -0.062^{***} (0.011) | -0.435^{***} (0.011) | 0.169^{**} (0.013) | 0.083^{***} (0.007) | $\begin{array}{c} 0.192^{***} \\ (0.007) \\ 0.007) \end{array}$ | $\begin{array}{c} 0.053^{***} \\ (0.014) \\ 0.023^{***} \end{array}$ | -0.118^{***} (0.016) |
| q_{iit-1} | 0.002^{**} | 0.002^{***} | 0.006^{***} | 0.003^{***} | 0.008^{***} | 0.002^{*} | 0.002^{***} | 0.006^{***} | 0.003^{**} | 0.005^{***} |

(0.016) 0.005^{***} (0.001)

(0.014) 0.003^{**} (0.001)

(0.001)

(0.001)

 $(0.013) \\ 0.002^{*}$ (0.001)

(0.011) 0.008^{***} (0.001)

(0.011) 0.003^{***} (0.001)

 $(0.011) \\ 0.002^{**}$ (0.001)

 q_{ijt-1}

(0.001)

(0.001)

 $_{
m 5,940}^{
m Yes}$ 0.044

 $\mathop{\mathrm{Yes}}_{5,940}$ 0.052

 $_{5,940}^{
m Yes}$ 0.310

 $\substack{\mathrm{Yes}\\5,940\\0.188}$

 $_{5,940}^{
m Yes}$ 0.107

 $\mathop{\mathrm{Yes}}_{5,940}$ 0.087

 $\mathop{\mathrm{Yes}}_{5,940}$ 0.035

 $\substack{\text{Yes}\\5,940\\0.298}$

 $_{5,940}^{\mathrm{Yes}}$ 0.185

 $_{
m 5,940}^{
m Yes}$ 0.112

Controls N

 R^2

Panel A. CF1

| om operating ry items <i>plus</i> n-utility and in Appendix respectively. | relations | 1 CF&CF2 | * 0.703*** | $*$ 0.763 *** | * 0.526*** | $* 0.638^{***}$ | $* 0.536^{***}$ | $* 0.744^{***}$ | $* 0.740^{***}$ | $* 0.670^{***}$ | $*$ 0.752 *** | $* 0.740^{***}$ | $* 0.759^{***}$ | $*$ 0.751 *** | * 0.705*** | * 0.709*** | * 0.734*** | * 0.783*** | * 0.712*** | | | | |
|-----------------------------------------------------------------------------------------------------------|-----------|----------|---------------|--------------------|---------------|-----------------|-----------------|-----------------|-----------------|-----------------|--------------------|-----------------|-----------------|--------------------|---------------|---------------|---------------|---------------|---------------|------------------------|------------------------------------------|---------|---------------|
| sh flow fr traordinau listed no e defined nt levels, | Corr | CF&CF | 0.768**: | $0.806^{**:}$ | $0.658^{**:}$ | $0.821^{**:}$ | $0.626^{**:}$ | $0.669^{**:}$ | 0.737^{**} | $0.654^{**:}$ | $0.741^{**:}$ | $0.683^{**:}$ | $0.646^{**:}$ | $0.667^{**:}$ | $0.638^{**:}$ | $0.483^{**:}$ | $0.435^{**:}$ | $0.623^{**:}$ | $0.666^{**:}$ | | | | |
| F is net ca e before ex consists of bles used ar l ten perce | | Std.Dev | 0.107^{***} | 0.166^{***} | 0.146^{***} | 0.110^{***} | 0.123^{***} | 0.149^{***} | 0.140^{***} | 0.143^{***} | 0.151^{***} | 0.115^{***} | 0.104^{***} | 0.107^{***} | 0.102^{***} | 0.108^{***} | 0.110^{***} | 0.105^{***} | 0.126^{***} | 0.100 | 0.026 | 20.63 | [0.00] |
| ash flow. C net incom he sample All variat ne, five, and | CF2 | Median | 0.171^{***} | 0.172^{***} | 0.177^{***} | 0.183^{***} | 0.202^{***} | 0.197^{***} | 0.213^{***} | 0.225^{***} | 0.207^{***} | 0.168^{***} | 0.166^{***} | 0.173^{***} | 0.174^{***} | 0.160^{***} | 0.156^{***} | 0.145^{***} | 0.178^{***} | 0.138 | 0.040 | 22.47 | [0.000] |
| proxies of $c_{\text{sets. CF2}}$ is sets. CF2 is $0 + \Delta E$. T 2000-2015. tce at the on | | Mean | 0.185^{***} | 0.199^{***} | 0.201^{***} | 0.185^{***} | 0.195^{***} | 0.218^{***} | 0.229^{***} | 0.235^{***} | 0.239^{***} | 0.185^{***} | 0.181^{***} | 0.188^{***} | 0.181^{***} | 0.177^{***} | 0.166^{***} | 0.158^{***} | 0.195^{***} | 0.156 | 0.039 | 20.00 | [0.000] |
| tween the I to-total ass to-total ass - Div $+ \Delta \Gamma$ the period is significan | | Std.Dev | 0.096^{***} | 0.157^{***} | 0.132^{***} | 0.096^{***} | 0.110^{***} | 0.142^{***} | 0.131^{***} | 0.161^{***} | 0.152^{***} | 0.123^{***} | 0.111^{***} | 0.113^{***} | 0.110^{***} | 0.122^{***} | 0.109^{***} | 0.121^{***} | 0.131^{***} | 0.100 | 0.031 | 23.66 | [0.000] |
| relations be spreciation- + Capex + <i>tream</i> over **, * indicat | CF1 | Median | 0.203^{***} | 0.198^{***} | 0.207^{***} | 0.215^{***} | 0.225^{***} | 0.232^{***} | 0.257^{***} | 0.250^{***} | 0.240^{***} | 0.201^{***} | 0.189^{***} | 0.189^{***} | 0.191^{***} | 0.173^{***} | 0.158^{***} | 0.149^{***} | 0.203^{***} | 0.138 | 0.065 | 32.02 | [0.000] |
| airwise cori ome $plus$ de = $\Delta Cash$ from $Datas$ triles. ***, | | Mean | 0.199^{***} | 0.236^{***} | 0.221^{***} | 0.220^{***} | 0.231^{***} | 0.255^{***} | 0.269^{***} | 0.274^{***} | 0.269^{***} | 0.199^{***} | 0.200^{***} | 0.200^{***} | 0.191^{***} | 0.167^{***} | 0.164^{***} | 0.156^{***} | 0.214^{***} | 0.156 | 0.058 | 27.10 | [0.000] |
| istics and p erating inco s. $\sum Uses_i$ ries drawn one percen | | Std.Dev | 0.087 | 0.140 | 0.102 | 0.082 | 0.097 | 0.098 | 0.107 | 0.114 | 0.123 | 0.099 | 0.088 | 0.091 | 0.083 | 0.088 | 0.100 | 0.091 | 0.100 | 0.180 | -0.080 | - 80.00 | [0.000] |
| mmary stat . CF1 is op -total asset rican count r and upper | CF | Median | 0.111 | 0.139 | 0.140 | 0.139 | 0.144 | 0.142 | 0.158 | 0.160 | 0.155 | 0.143 | 0.136 | 0.137 | 0.137 | 0.120 | 0.123 | 0.116 | 0.138 | 0.175 | -0.037 | - 26.81 | [0.000] |
| ne-series su total assets lividends-to selected Af at the lowe | | Mean | 0.137 | 0.173 | 0.167 | 0.149 | 0.165 | 0.164 | 0.176 | 0.180 | 0.186 | 0.160 | 0.147 | 0.149 | 0.141 | 0.137 | 0.144 | 0.136 | 0.156 | 0.194 | -0.038 | - 24.36 | [0.000] |
| presents tin NOCF)-to- on <i>minus</i> c ial firms in winsorised | | Ν | 171 | 231 | 278 | 310 | 331 | 340 | 365 | 392 | 401 | 444 | 487 | 508 | 477 | 447 | 412 | 346 | 5,940 | | $\sum \mathrm{Uses}_i$ | (%) | ue |
| The table activities (depreciatic non-financ A and are | | Year | 2000 | 2001 | 2002 | 2003 | 2004 | 2005 | 2006 | 2007 | 2008 | 2009 | 2010 | 2011 | 2012 | 2013 | 2014 | 2015 | CF | $\sum \mathrm{Uses}_i$ | $\overline{\mathrm{Diff}}=\mathrm{CF}$ - | Diff/CF | Diff p -vai |

Appendix E The evolution of the proxies of cash flow

68

Online Appendices

Not For Publication

Appendix A1 Basic statistics by country

The table presents the summary statistics of the key variables by country. The sample consists of listed non-utility and non-financial firms in selected African countries drawn from *Datastream* over the period 2000–2015. All variables used are defined in Appendix A and are winsorised at the lower and upper one percentiles. ***, **, * indicate significance at the one, five, and ten percent levels, respectively.

| FIC | Variables | CF | $\Delta Cash$ | Capex | Div | $\Delta \mathrm{Debt}$ | $\Delta Equity$ |
|-------------|-----------------------------------------|---------------------------------------------------------------------------|------------------------------------------------------------------------|-----------------------------------------------|----------------------------------------------------------------------------|-------------------------------------------------------------------------|--------------------------------------|
| Egypt | Mean Median Std.Dev Trend N | $\begin{array}{c} 0.014 \\ 0.005 \\ 0.084 \\ 0.043 \\ 483 \end{array}$ | 0.056 0.026 0.076 -0.489*** | 0.085 0.064 0.080 -0.031 Firms | $\begin{array}{c} 0.007 \\ 0.000 \\ 0.074 \\ 0.276^* \\ 60 \end{array}$ | -0.071 -0.066 0.112 -0.085 | 0.162 0.132 0.117 -0.616** |
| Ghana | Mean Median Std.Dev Trend N | $\begin{array}{c} 0.042 \\ 0.009 \\ 0.105 \\ 0.104 \\ 46 \end{array}$ | $\begin{array}{c} 0.118 \\ 0.100 \\ 0.092 \\ 0.559 \end{array}$ | 0.056 0.044 0.055 -0.037 Firms | -0.001 0.000 0.099 1.523** 7 | -0.049 -0.056 0.139 -0.741 | 0.214 0.163 0.156 -0.203 |
| Ivory Coast | Mean Median Std.Dev Trend N | $\begin{array}{c} -0.012 \\ 0.003 \\ 0.061 \\ 0.667 \\ 48 \end{array}$ | $\begin{array}{c} 0.065\\ 0.049\\ 0.074\\ 1.143^{***}\end{array}$ | 0.122 0.115 0.092 -1.090** Firms | $\begin{array}{c} 0.022 \\ 0.000 \\ 0.067 \\ 0.714^{***} \\ 7 \end{array}$ | -0.023 -0.052 0.144 -0.721 | 0.203 0.199 0.120 -0.607 |
| Kenya | Mean Median Std.Dev Trend N | $\begin{array}{c} 0.013 \\ 0.011 \\ 0.066 \\ 0.059 \\ 145 \end{array}$ | 0.089 0.071 0.068 -0.093 | 0.070 0.035 0.066 -0.189 Firms | $\begin{array}{c} 0.019 \\ 0.000 \\ 0.088 \\ 0.305 \\ 19 \end{array}$ | -0.079 -0.073 0.117 -0.288* | 0.168 0.138 0.107 -0.368 |
| Morocco | Mean Median Std.Dev Trend N | -0.004 0.001 0.059 0.819*** 138 | $\begin{array}{c} 0.096 \\ 0.100 \\ 0.065 \\ -0.628^{***} \end{array}$ | 0.113 0.112 0.083 -1.039*** Firms | $\begin{array}{c} 0.017\\ 0.002\\ 0.062\\ 0.244^{*}\\ 20 \end{array}$ | -0.097 -0.069 0.112 1.315*** | 0.207 0.201 0.127 -1.520*** |
| Nigeria | Mean Median Std.Dev Trend N | $\begin{array}{c} 0.019 \\ 0.006 \\ 0.101 \\ -0.634^* \\ 105 \end{array}$ | 0.134 0.119 0.103 -0.061 | 0.079 0.055 0.074 -0.582** Firms | $\begin{array}{c} 0.027 \\ 0.000 \\ 0.118 \\ 0.714 \\ 14 \end{array}$ | -0.080 -0.065 0.103 -0.228 | 0.211 0.192 0.111 -1.065*** |
| S. Africa | Mean Median Std.Dev Trend N | $\begin{array}{c} 0.019 \\ 0.008 \\ 0.079 \\ -0.034 \\ 4,798 \end{array}$ | $\begin{array}{c} 0.079 \\ 0.065 \\ 0.059 \\ -0.137^{***} \end{array}$ | 0.048 0.030 0.063 0.122*** Firms | 0.021 0.001 0.085 0.124*** 485 | $\begin{array}{c} -0.058 \\ -0.056 \\ 0.110 \\ 0.173^{***} \end{array}$ | 0.152 0.136 0.096 -0.255*** |
| Tunisia | Mean Median Std.Dev Trend N | $\begin{array}{c} 0.010 \\ 0.011 \\ 0.053 \\ 0.122 \\ 177 \end{array}$ | 0.080 0.076 0.047 -0.437* | 0.061 0.061 0.036 -0.215** Firms | $\begin{array}{c} 0.005 \\ 0.000 \\ 0.057 \\ 0.038 \\ 27 \end{array}$ | $\begin{array}{c} -0.072 \\ -0.074 \\ 0.041 \\ 0.101 \end{array}$ | 0.147 0.145 0.070 -0.521 |